FANUC Robot series

R-30*i*B/R-30*i*B Mate/ R-30*i*B Plus/ R-30*i*B Mate Plus CONTROLLER

OPERATOR'S MANUAL (Collaborative Robot Function)

MAROBCLBR09151E REV B

©2017 FANUC Robotics America Corporation All Rights Reserved.

This publication contains proprietary information of FANUC America Corporation furnished for customer use only. No other uses are authorized without the express written permission of FANUC America Corporation.

> FANUC America Corporation 3900 W. Hamlin Road Rochester Hills, Michigan 48309–325

> > B-83744EN/02

Copyrights and Trademarks

This new publication contains proprietary information of FANUC America Corporation furnished for customer use only. No other uses are authorized without the express written permission of FANUC America Corporation.

The descriptions and specifications contained in this manual were in effect at the time this manual was approved for printing. FANUC America Corporation, hereinafter referred to as FANUC, reserves the right to discontinue models at any time or to change specifications or design without notice and without incurring obligations.

FANUC manuals present descriptions, specifications, drawings, schematics, bills of material, parts, connections and/or procedures for installing, disassembling, connecting, operating and programming FANUC products and/or systems. Such systems consist of robots, extended axes, robot controllers, application software, the KAREL® programming language, INSIGHT® vision equipment, and special tools.

FANUC recommends that only persons who have been trained in one or more approved FANUC Training Course(s) be permitted to install, operate, use, perform procedures on, repair, and/or maintain FANUC products and/or systems and their respective components. Approved training necessitates that the courses selected be relevant to the type of system installed and application performed at the customer site.

A WARNING

This equipment generates, uses, and can radiate radiofrequency energy and if not installed and used in accordance with the instruction manual, may cause interference to radio communications. As temporarily permitted by regulation, it has not been tested for compliance with the limits for Class A computing devices pursuant to Subpart J and Part 15 of FCC Rules, which are designed to provide reasonable protection against such interference. Operation of the equipment in a residential area is likely to cause interference, in which case the user, at his own expense, will be required to take whatever measure may be required to correct the interference. FANUC conducts courses on its systems and products on a regularly scheduled basis at the company's world headquarters in Rochester Hills, Michigan. For additional information contact

FANUC America Corporation Training Department 3900 W. Hamlin Road Rochester Hills, Michigan 48309-3253 www.fanucamerica.com

For customer assistance, including Technical Support, Service, Parts & Part Repair, and Marketing Requests, contact the Customer Resource Center, 24 hours a day, at 1-800-47-ROBOT (1-800-477-6268). International customers should call 011-1-248-377-7159.

Send your comments and suggestions about this manual to: product.documentation@fanucamerica.com

Copyright © 2017 by FANUC America Corporation All Rights Reserved

The information illustrated or contained herein is not to be reproduced, copied, downloaded, translated into another language, published in any physical or electronic format, including internet, or transmitted in whole or in part in any way without the prior written consent of FANUC America Corporation.

AccuStat®, ArcTool®, *i*RVision®, KAREL®, PaintTool®, PalletTool®, SOCKETS®, SpotTool®, SpotWorks®, and TorchMate® are Registered Trademarks of FANUC.

FANUC reserves all proprietary rights, including but not limited to trademark and trade name rights, in the following names:

AccuAirTM, AccuCalTM, AccuChopTM, AccuFlowTM, AccuPathTM, AccuSealTM, ARC MateTM, ARC Mate Sr.TM, ARC Mate System 1TM, ARC Mate System 2TM, ARC Mate System 3TM, ARC Mate System 4TM, ARC Mate System 5TM, ARCWorks ProTM, AssistToolTM, AutoNormalTM, AutoTCPTM, BellToolTM, BODYWorksTM, Cal MateTM, Cell FinderTM, Center FinderTM, Clean WallTM, DualARMTM, LR ToolTM, MIG EyeTM, MotionPartsTM, MultiARMTM, NoBotsTM, Paint StickTM, PaintProTM, PaintTool 100TM, PAINTWorksTM, PAINTWorks IITM, PAINTWorks IIITM, PalletMateTM, PalletMate PCTM, PalletTool PCTM, PayloadIDTM, RecipToolTM, RemovalToolTM, Robo ChopTM, Robo SprayTM, S-420iTM, S-430iTM, ShapeGenTM, SoftFloatTM, SOFT PARTSTM, SpotTool+TM, SR MateTM, SR ShotToolTM, SureWeldTM, SYSTEM R-J2 ControllerTM, SYSTEM R-J3 Controller[™], SYSTEM R-J3*i*B Controller[™], SYSTEM R-J3*i*C Controller[™], SYSTEM R-30*i*A Controller[™], SYSTEM R-30*i*A Mate Controller[™], SYSTEM R-30*i*B Controller[™], SYSTEM R-30*i*B Mate Controller[™], SYSTEM R-30*i*B Plus ControllerTM. SYSTEM R-30*i*B Mate Plus ControllerTM. TCP MateTM. TorchMateTM, TripleARMTM, TurboMoveTM, visLOCTM, visPRO-3DTM, visTRACTM, WebServerTM, WebTPTM, and YagToolTM.

©FANUC CORPORATION 2017

- No part of this manual may be reproduced in any form.
- All specifications and designs are subject to change without notice.

Patents

One or more of the following U.S. patents might be related to the FANUC products described in this manual.

FANUC America Corporation Patent List

 $\begin{array}{l} 4,630,567\ 4,639,878\ 4,707,647\ 4,708,175\ 4,708,580\ 4,942,539\ 4,984,745\\ 5,238,029\ 5,239,739\ 5,272,805\ 5,293,107\ 5,293,911\ 5,331,264\ 5,367,944\\ 5,373,221\ 5,421,218\ 5,434,489\ 5,644,898\ 5,670,202\ 5,696,687\ 5,737,218\\ 5,823,389\ 5,853,027\ 5,887,800\ 5,941,679\ 5,959,425\ 5,987,726\ 6,059,092\\ 6,064,168\ 6,070,109\ 6,086,294\ 6,122,062\ 6,147,323\ 6,204,620\ 6,243,621\\ 6,253,799\ 6,285,920\ 6,313,595\ 6,325,302\ 6,345,818\ 6,356,807\ 6,360,143\\ 6,378,190\ 6,385,508\ 6,425,177\ 6,477,913\ 6,490,369\ 6,518,980\ 6,540,104\\ 6,541,757\ 6,560,513\ 6,569,258\ 6,612,449\ 6,703,079\ 6,705,361\ 6,726,773\\ 6,768,078\ 6,845,295\ 6,945,483\ 7,149,606\ 7,149,606\ 7,211,978\ 7,266,422\\ 7,399,363\end{array}$

FANUC CORPORATION Patent List

4,571,694 4,626,756 4,700,118 4,706,001 4,728,872 4,732,526 4,742,207 4,835,362 4,894,596 4,899,095 4,920,248 4,931,617 4,934,504 4,956,594 4,967,125 4,969,109 4,970,370 4,970,448 4,979,127 5,004,968 5,006,035 5,008,834 5,063,281 5,066,847 5,066,902 5,093,552 5,107,716 5,111,019 5,130,515 5,136,223 5,151,608 5,170,109 5,189,351 5,267,483 5,274,360 5,292,066 5,300,868 5,304,906 5,313,563 5,319,443 5,325,467 5,327,057 5,329,469 5,333,242 5,337,148 5,371,452 5,375,480 5,418,441 5,432,316 5,440,213 5,442,155 5,444,612 5,449,875 5,451,850 5,461,478 5,463,297 5,467,003 5,471,312 5,479,078 5,485,389 5,485,552 5,486,679 5,489,758 5,493,192 5,504,766 5,511,007 5,520,062 5,528,013 5,532,924 5,548,194 5,552,687 5,558,196 5,561,742 5,570,187 5,570,190 5,572,103 5,581,167 5,582,750 5,587,635 5,600,759 5,608,299 5,608,618 5,624,588 5,630,955 5,637,969 5,639,204 5,641,415 5,650,078 5,658,121 5,668,628 5,687,295 5,691,615 5,698,121 5,708,342 5,715,375 5,719,479 5,727,132 5,742,138 5,742,144 5,748,854 5,749,058 5,760,560 5,773,950 5,783,922 5,799,135 5,812,408 5,841,257 5,845,053 5,872,894 5,887,122 5,911,892 5,912,540 5,920,678 5,937,143 5,980,082 5,983,744 5,987,591 5,988,850 6,023,044 6,032,086 6,040,554 6,059,169 6,088,628 6,097,169 6,114,824 6,124,693 6,140,788 6,141,863 6,157,155 6,160,324 6,163,124 6,177,650 6,180,898 6,181,096 6,188,194 6,208,105 6,212,444 6,219,583 6,226,181 6,236,011 6,236,896 6,250,174 6,278,902 6,279,413 6,285,921 6,298,283 6,321,139 6,324,443 6,328,523 6,330,493 6,340,875 6,356,671 6,377,869 6,382,012 6,384,371 6,396,030 6,414,711 6,424,883 6,431,018 6,434,448 6,445,979 6,459,958 6,463,358 6,484,067 6,486,629 6,507,165 6,654,666 6,665,588 6,680,461 6,696,810 6,728,417 6,763,284 6,772,493 6,845,296 6,853,881 6,888,089 6,898,486 6,917,837 6,928,337 6,965,091 6,970,802 7,038,165 7,069,808 7,084,900 7,092,791 7,133,747 7,143,100 7,149,602 7,131,848 7,161,321 7,171,041 7,174,234 7,173,213 7,177,722 7,177,439 7,181,294 7,181,313 7,280,687 7,283,661 7,291,806 7,299,713 7,315,650 7,324,873 7,328,083 7,330,777 7,333,879 7,355,725 7,359,817 7,373,220 7,376,488 7,386,367 7,464,623 7,447,615 7,445,260 7,474,939 7,486,816 7,495,192 7,501,778 7,502,504 7,508,155 7,512,459 7,525,273 7,526,121

Conventions

Information appearing under the "WARNING" caption concerns the protection of personnel. It is boxed and bolded to set it apart from the surrounding text.

Information appearing under the "CAUTION" caption concerns the protection of equipment, software, and data. It is boxed and bolded to set it apart from the surrounding text.

Note Information appearing next to NOTE concerns related information or useful hints.

Original Instructions

Thank you very much for purchasing FANUC Collaborative Robot.

Before using the Collaborative robot, be sure to read MECHANICAL UNIT OPERATOR'S MANUAL and understand the content.

- No part of this manual may be reproduced in any form.
- The appearance and specifications of this product are subject to change without notice.

The products in this manual are controlled based on Japan's "Foreign Exchange and Foreign Trade Law". The export from Japan may be subject to an export license by the government of Japan. Further, re-export to another country may be subject to the license of the government of the country from where the product is re-exported. Furthermore, the product may also be controlled by re-export regulations of the United States government. Should you wish to export or re-export these products, please contact FANUC for advice.

In this manual, we endeavor to include all pertinent matters. There are, however, a very large number of operations that must not or cannot be performed, and if the manual contained them all, it would be enormous in volume. It is, therefore, requested to assume that any operations that are not explicitly described as being possible are "not possible".

SAFETY PRECAUTIONS

Before using the FANUC collaborative robot, be sure to read MECHANICAL UNIT OPERATOR'S MANUAL to become familiar with those contents.

DEFINITION OF SAFETY NOTATIONS

To ensure the safety of users and prevent damage to the machine, this manual indicates each precaution on safety with "WARNING" or "CAUTION" according to its severity. Supplementary information is indicated by "NOTE". Please read each "WARNING", "CAUTION" and "NOTE" before using the robots.

Symbol	Definitions
	Used if hazard resulting in the death or serious injury of the user will be expected to
	Used if a hazard resulting in the minor or moderate injury of the user, or equipment damage may be expected to occur if he or she fails to follow the approved procedure.
NOTE	Used if a supplementary explanation not related to any of WARNING and CAUTION is to be indicated.

• Check this manual thoroughly, and keep it handy for the future reference.

PREFACE

The coexistence of human and robot will make a difference in production site from the point of view of productivity, efficiency and so on. However, for coexistence of human and robot, adequate risk assessment for the whole the robot system is necessary to verify the human safety.

Collaborative robot is befitted for this purpose, because the collaborative robot has the contact stop function to stop when the external force exceeds the limit. This function is useful for design of the robot system.



The collaborative robot is certified to meet the requirements of International Standard ISO 10218-1 by a notarized body.

The contact stop function is certified to meet the requirements of International ISO 13849-1 for Category 3, PL (Performance Level) d by a notarized body.

In order to design the collaborative robot system, it is necessary to understand the collaborative robot function. This manual describes the collaborative robot function use.

RELATED MANUALS

For the FANUC Robot series, the following manuals are available:

R-30 <i>i</i> B,	OPERATOR'S MANUAL	Intended readers :
R-30iB Mate,	Basic Operation	Operator, programmer, maintenance engineer, system
R-30 <i>i</i> B Plus,	B-83284EN	designer
R-30 <i>i</i> B Mate Plus	Alarm Code List	Topics :
controller	B-83284EN-1	Robot functions, operations, programming, setup,
	Optional Function	interfaces, alarms
	B-83284EN-2	Use :
	Dual Check Safety Function	Robot operation, teaching, system design
	B-83184EN	
	MAINTENANCE MANUAL	Intended readers :
	B-83525EN	Maintenance engineer, system designer
		Topics :
		Installation, start-up, connection, maintenance
		Use :
		Installation, start-up, connection, maintenance
FANUC Robot	OPERATOR'S MANUAL	Intended readers:
CR-4 <i>i</i> A,	B-83774EN	System designer, Maintenance engineer
CR-7 <i>i</i> A,		Topics:
CR-7 <i>i</i> A/L		Installation, connection to controller, maintenance
Mechanical unit		Use:
		Installation, start-up, connection, maintenance
FANUC Robot	OPERATOR'S MANUAL	Intended readers:
CR-35 <i>i</i> A	B-83734EN	System designer, Maintenance engineer
Mechanical unit		Topics:
		Installation, connection to controller, maintenance
		Use:
		Installation, start-up, connection, maintenance

This manual uses following terms.

Name	Terms in this manual			
Robot mechanical unit	Mechanical unit			

TABLE OF CONTENTS

SAF	ETY P	RECAUTIONS	;-1
PRE	FACE	β)-1
1	ABST 1.1 1.2 1.3	RACT RESTRICTION OF COLLABORATIVE ROBOT SOFTWARE OPTION FOR COLLABORATIVE ROBOT SETUP STEPS	 1 1 2
2	SETTI 2.1 2.2 2.3	NG UP SETTING SAFE I/O DEVICE. SETTING COLLABORATIVE ROBOT FUNCTION 2.2.1 Verify Payload Setup OUTPUT SIGNAL OF COLLABORATIVE ROBOT STATUS	3 4 9 11
3	PAYL 3.1	OAD CONFIRMATION OPERATION AFTER POWER ON Payload Confirmation without Teach Pendant	13 14
4	SPEC 4.1 4.2	IFICATION OF COLLABORATIVE FUNCTION CONTACT STOP FUNCTION 4.1.1 Contact Stop in Teach Mode COLLABORATIVE ROBOT MONITOR 4.2.1 Auto Status Check 4.2.2 Payload Monitor 4.2.3 Collaborative Robot Monitor Screen	15 .16 .16 16 16 17
	4.3 4.4	RETREAT AFTER CONTACT STOP PUSH TO ESCAPE	19 20
5	PROG 5.1 5.2	RAMMING PICKING WORKPIECE 5.1.1 Simple Picking Program 5.1.2 Limiting Picking Area DISABLING CONTACT STOP IN PROGRAM	21 .21 .21 .24 28
6	PERIC	DDICAL CHECK	35
7	UTILII 7.1 7.2 7.3 7.4 7.5	AUTO RESUME FUNCTION HIGH SPEED MODE LED AND BUTTON ON ROBOT ARM OPERATION LOG BOOK Variable Payload Compensation function	37 38 41 42 43 44 45

8	iHM		COLLABORATIVE ROBOT	
•	8.1	OVER	RVIEW OF THE HOME SCREEN	
	8.2	OPER	RATIONS RELATED TO SETUP	
		8.2.1	Collaborative Setup Guide	
	8.3	OPER	RATIONS RELATED TO TEACH	51
		8.3.1	Create Program	
		8.3.2	Pick & Place with Light Work	
		8.3.3	Pick & Place with Heavy Work	
	8.4	Collab	porative Monitor	
		8.4.1	Collaborative Robot Monitor Screen	

APPENDIX

Α	TRO	UBLESHOOTING	59
	A.1	INDELIBLE ALARM	59
	A.2	STOP DURING MOVE	60
	A.3	STOP DURING PERIODICAL CHECK	60

1 ABSTRACT

1.1 RESTRICTION OF COLLABORATIVE ROBOT

- Constant path function is needed for the collaborative robot function. So you can NOT use any option unsupported by Constant path function.
- Circular motion and Circle arc motion are not supported.
- Payload Identification is not supported.

1.2 SOFTWARE OPTION FOR COLLABORATIVE ROBOT

Robot library software of Collaborative robot is sufficient for the simple collaborative system. For more advanced system as followings, DCS function is required. Refer to Chapter 5.

- Limit the area for picking or placing a workpiece
- Disable the contact stop function in specified area



Recommended DCS functions

- J567 DCS Position/Speed check function
- J568 DCS Safe I/O connection function

DCS Joint Position Check function is installed as standard. Refer to Dual Check Safety Function OPERATOR'S MANUAL (B-83184EN) for detail of this function.

1.3 SETUP STEPS

To use the collaborative robot, you must do the followings. You can use the other functions except for the collaborative robot function as same manner as the standard FANUC robot. Refer to each function manual.

- 1 Set up the collaborative robot function. Refer to Chapter 2. For more detail information of the collaborative robot function, refer to Chapter 4.
- 2 Perform the payload confirmation operation after every power on. Refer to Chapter 3.
- 3 Teach production programs. Refer to Chapter 5 of this manual and OPERATOR'S MANUAL (Basic Operation) (B-83284EN), Chapter of PROGRAMMING.
- 4 Make periodical check program of force sensor. Refer to Chapter 6.
- 5 Run production with above programs. Refer to OPERATOR'S MANUAL (Basic Operation) (B-83284EN), Chapter of EXECUTING A PROGRAM.
- 6 Set up the special functions of the collaborative robot. Refer to Chapter 7.

If you have any problem of the collaborative robot, refer to Appendix A.

2 SETTING UP

This chapter describes how to set up the collaborative robot. By default, SAFE I/O device is set correctly. Set up after Section 2.2.

2.1 SETTING SAFE I/O DEVICE

Collaborative robot has a built-force sensor, and this sensor detects a external force to realize the contact stop function. By default, the four "Force Sensor" devices are assigned to Device 2 to 5. These devices are needed for the collaborative robot function. If the safe I/O device configuration is changed, the following operation is needed.

Initialize Safe I/O device

- 1 Press [MENU] key.
- 2 Select SYSTEM.
- 3 Press F1, [TYPE].
- 4 Select DCS. If neither DCS top screen nor Safe I/O device screen, press [PREV] key, and select Safe I/O device.
- 5 Press F2, [INIT]. The message "Do you want to change setting?" is displayed.
- 6 Press F4, [YES].
- 7 Press F2, [INIT] again. The message "Initialize device configuration?" is displayed.
- 8 Press F4, [YES].
- 9 Make sure that the types of Device 2 to 5. The following screen will be displayed.

DCS	5							
	Safe I/	0 device	3					
	1 Saf	e I/O d	evice	: I/O	conr	lect	OK	
	2 SFD	0 Pulse	checl	C: ENA	BLE(]	PL e)	OK	
	Configu	iration					status	
		- Device	1					
	3 Тур	e: E-sto	p boa	ard			OK	
	4 Rob	ot numbe	r: 1				OK	
	5 Inp	ut: SP	[[1	- 8](PL	e)	OK	
	6 Out	put: SP	0[1	- 8](PL	e)	OK	
		- Device	2					
	7 Тур	e: Force	Sens	sor			CHGD	
	8 Rob	ot numbe	r: 0				OK	
	9 Inp	ut: SP	τ[Ο	- 0](PL	e)	OK	
	10 Out	put: SP	0[0	- 0](PL	e)	OK	
		- Device	3					
	11 Typ	e: Force	Sen	sor			CHGD	
	12 Rob	ot numbe	er: 0				OK	
	13 Inp	out: SP	I[0	- 0](PL	e)	OK	
	14 Out	put: SP	0]0	- 0](PL	e)	OK	
		- Device	4					
	15 Typ	e: Force	Sen	sor			CHGD	
	16 Rob	ot numbe	er: 0	_			OK	
	17 In <u>r</u>	out: SP	I[0	- 0](PL	e)	OK	
	18 Out	put: SP	0]0	- 0](PL	e)	OK	
		- Device	5					
	19 Typ	e: Force	Sen	sor			CHGD	
	20 Rob	ot numbe	er: 0	_			OK	
	21 In <u>r</u>	out: SP	I[0	- 0](PL	e)	OK	
	22 Out	put: SP	0[0	- 0](PL	e)	OK	
	·	<u> </u>	-		1			I
	[TYPE]	INIT					UNDO	

10 Apply DCS parameter. Refer to Dual Check Safety Function OPERATOR'S MANUAL (B-83184EN), Chapter of OVERVIEW, Section of APPLY TO DCS PARAMETER.

2.2 SETTING COLLABORATIVE ROBOT FUNCTION

This section introduces the setting of the collaborative robot function. In the collaborative robot screen, you can set up the collaborative robot function.

Collaborative robot screen

- 1 Press the [MENU] key.
- 2 Select SYSTEM.
- 3 Press F1, [TYPE].
- 4 Select DCS.
- 5 If neither DCS top screen nor Collaborative robot screen, press the [PREV] key, and select Collaborative robot. The following screen will be displayed. Some items is not displayed depending on the software version.

DCC	,					
DCS						
	COLLADO	rative ro	DOT			
	Contact	stop sta	tus ST	OP		
	Enable/	Disable	EN	ABLE	OK	
	Force s	ensor				
	Seri	al number	•	0	OK	
	Group:			1	OK	
	Payload	setup:	<d< th=""><th>ETAIL></th><th>OK</th><th></th></d<>	ETAIL>	OK	
	Active	Payload n	umber: No	. 1 [****	******]	
	Externa	l force L	imit / D:	sabling i	input	
	Currei	nt: (0.0	-	-	
	Limit	1: 150	.00[N]	-[0]	OK	
	Limit	2: 0	.00[N]	-[0]	OK	
	Limit	3: 0	.00[N]	-[0]	OK	
	Limit	4: 0	.00[N]	- [0]	OK	
	Fedar	300	00[N]		OK	
	Басар	z. 500	.00[N]		OK	
	[TYPE]	CONFIRM	PEAKCLR		UNDO	

You can set up the collaborative function in this screen.

For configurable items, OK/CHGD/PEND is displayed in the right.

OK : Setting parameter and DCS parameter are the same.

CHGD : Setting parameter is changed, but not applied to DCS parameter.

PEND : Setting parameter is changed and applied to DCS parameter, but controller power has not been cycled.

When you change DCS parameter setting, "Apply to the DCS parameter" is necessary. Refer to Dual Check Safety Function OPERATOR'S MANUAL (B-83184EN), Chapter of OVERVIEW, Section of APPLY TO DCS PARAMETER.

DCS	DCS							
	Collabo	rative rol						
	Contact	stop sta	TOP					
	Enable/	Disable	E	NABLI	3	OK		
	Force s							
	Seri	al number.	:		0	OK		
	Group:				1	OK		
	Payload setup: <i< td=""><td>IL></td><td>OK</td><td></td></i<>				IL>	OK		
	Active Payload number: No.				. [****	*****]		
	External force Limit / Disabl				oling i	nput		
	Currei	nt: C	0.0					
	Limit	1: 150	.00[N]	[0]	OK		
	Limit	2: 0.	.00[N] -	[0]	OK		
	Limit	3: 0.	.00[N] -	[0]	OK		
	Limit	4: 0.	.00[N] -	[0]	OK		
	Escape	e: 300	.00[N]			OK		
	[TYPE]	CONFIRM	PEAKCL	R		UNDO		

Contact stop status

The status of the contact stop function is displayed. Refer to Section 2.3.

Enable/Disable (configurable)

Enabling or disabling the contact stop function.

DISABLE

The contact stop function is disabled. In this setting, the robot is same as a standard robot. You can move the robot only in T1 mode. In T2 or AUTO mode, "SYST-343 Enable Contact Stop function" occurs, and you cannot move the robot.

ENABLE

The contact stop function is enabled.

ENBL (Shift+Reset)

Contact stop is always enabled in AUTO mode. When you hold down the [SHIFT] key and press the [RESET] key in T1 or T2 mode and the contact stop status is STOP (Refer to Section 2.3), the contact stop function is disabled until the [SHIFT] key is released or the safety is confirmed.

Refer to Subsection 4.1.1 for the detail of the difference between ENABLE and ENBL(Shift+Reset).

When "ENBL (Shift+Reset)" is selected, the contact stop function might be disabled when you hold down the [SHIFT] key and press the [RESET] key in T1 or T2 mode. If "ENBL (Shift+Reset)" is used, adequate risk assessment for the whole the robot system is necessary to verify that the contact stop can be disabled while Shift+Reset is pressed in T1 or T2 mode.

Force Sensor Serial number

The serial number of the force sensor is displayed.

Group (configurable)

Set the group number of the collaborative robot.

Payload setup

You can see current payload setup. Refer to Subsection 2.2.1.

Active payload number

Current payload number is displayed and not configurable in this screen. You can select the payload number in "SYSTEM > Motion" screen or by PAYLOAD instruction in program. Refer to OPERATOR'S MANUAL (Basic Operation) (B-83284EN).

External force (N)

Current

You can see the current external force. This item moved to Force monitor in the software version 7DC3/30 or later.

Limit 1 to 4 (configurable)

Set the external force limit 1 to 4 between 0[N] and 150[N], inclusive. 0[N] means disable and you cannot set the limit 1 to 0[N].

When the contact stop function is enabled, "@" is displayed as following.

Limit 1	1:	150.00[N]	[0]	OK
@Limit 2	2:	90.00[N]	[0]	OK
Limit 3	3:	0.00[N]	[0]	OK
Limit 4	4:	0.00[N]	[0]	OK

This means the active force limit. The active force limit is minimum limit in available limits. When the external force exceeds the active force limit, the robot will stop. To disable the force limit, set the limit to 0[N] or the corresponding disabling input to ON. You can disable the force limit to change the active force limit dynamically in a program. In the example of this screen, the limit 1 and 2 are enabled, then @ is attached to the minimum limit 2. The limit 3 and 4 is set 0[N], then they are disabled.

Escape (configurable)

Set the external force limit for Push To Escape between 0[N] and 300[N], inclusive. In Push To Escape, if the external force exceeds this limit, the robot is stopped. This means that the robot does not escape when you push with a force more than the force limit.

Disabling input (configurable)

Set the Safe I/O to disable each force limit. When the disabling input is ON, the corresponding force limit is disabled. When the all limits are disabled, the contact stop function is disabled.

@Limit 1:	150.00[N] SIR[1]	OK
Limit 2:	90.00[N] SIR[2]	OK
Limit 3:	0.00[N][0]	OK
Limit 4:	0.00[N][0]	OK

In the example of this screen, SIR[2] is ON to disable the limit 2. The limit 1 is only enabled, then @ is attached to the limit 1.

DCS						
Collabo	rative ro	bot				
Force M	onitor:					
	Resulta	int 1	x	Y	z	
Curr:	0.00	[N]				
	0%	5 C)%	0%	0%	
Peak:	0%	s 0)%	0%	0%	
Outpu	t:					
Curi	GO[()]GO[0] (GO[0]	GO[0]	
Warni	ng: 0º	8				
Payload	Monitor:					
FORCE	Resulta	nt 1	ĸ	Y	Z	
Curr:	0%	s 0)%	0%	0%	
Comp:		0	%	0%	0%	
Peak:	0%	s 0)%	0%	0%	
Outpu	t:					
Curi	GO[0] GO[0]	GO[0]	GO[0]	
Com	?:	GO [0]	GO[0]	GO[0]	
Warni	ng: 50	8				
MOMENT	Result	ant	х	Y	Z	
Curr:	0%	s 0)%	0%	0%	
Comp:		0	8	0%	0%	
Peak:	0%	s 0)%	0%	0%	
Outpu	t:					
Curi	GO[0] GO[0]	GO[0]	GO[0]	
Com	?:	GO [0]	GO[0]	GO[0]	
Warni	ng: 50	%				
Warnin	g in DSBL	:		DISABLE	6	
[TYPE]	CONFIRM	PEAKCI	LR		UNDO	

Force Monitor/Payload Monitor

These items is displayed in the software version 7DC3/30 or later. An external force and a load applied to the robot is displayed. You can configure the warning posted when the monitored value exceeds specified value, and output monitored value. Refer to Subsection 4.2.3.

DCS									
	Collaborative robot								
	Payload	change d	istance	(mm)					
		Current	+Limit	-Limit					
	X:	0.00	0.00	0.00	OK				
	Y:	0.00	0.00	0.00	OK				
	z:	0.00	0.00	0.00	OK				
	Rotat	ion: Curr	ent	Limit					
		0.00	[deg]	0.20[deg]	OK				
	Enabl	ing input:	: ON	[0]	OK				
	Passwor	d for CON	FIRM:	ENABLE	OK				
	CONFIRM	by DI:		DISABLE	OK				
	CONFI	RM input:		DI[0]					
	Payloa	ad No. out	put:	GO[0]					
	Speed 1	imit: 250	0.00 [mm	/s]	OK				
	Use Pay	load Comp	:	ENABLE	OK				
	[TYPE]	CONFIRM	PEAKCLF	2	UNDO				

Payload change distance

Current

You can see the current distance and the rotation angle of the posture from a payload change position. The rotation angle of the posture is displayed in the software version 7DC3/30 or later. When the payload changing output is OFF (Refer to Section 2.3), this value is 0.

+Limit, -Limit, Rotation Limit (configurable)

When the payload change distance is enabled, the active payload number is changed, then the contact stop function will switch to disable. If the face plate position is changed more than the payload change distance, the contact stop function switches to enable immediately. The payload change distance can be set for each direction individually, +X, -X, +Y, -Y, +Z and -Z of the world coordinate system. Rotation Limit is displayed in the software version 7DC3/30 or later. Rotation Limit is one value without distinction of direction. This is useful for the collaborative robot to pick or place a workpiece. Refer to Chapter 5.

Enabling input (configurable)

Set the Safe I/O to enable the payload change distance. When the enabling input is ON, the payload change distance is enabled. When the Safe I/O is OFF, the payload change distance is disabled, and STOP state (Refer to Section 2.3) will be initiated immediately when the actual payload number is changed. Default setting is ON[0]. It means the payload change distance is always enabled.

Password for CONFIRM (configurable)

After power on the controller, the payload confirmation operation must be performed to clear the alarm. The collaborative robot does not move until the payload confirmation is performed. When Password for CONFIRM is ENABLE, DCS password is required for the payload confirmation operation. Default setting is ENABLE. Refer to Chapter 3.

CONFIRM by DI (configurable)

This item is displayed in the software version 7DC3/30 or later. You can perform the payload confirmation operation with some kind of terminal without other than the teach pendant. Refer to Chapter 3.

Speed limit (configurable)

This item is displayed in the software version 7DC3/28 or later. When the contact stop is enable, the speed of the collaborative robot exceeds the speed limit, "SYST-323 Collaborative speed limit" will be posted, and the robot will stop.

If the speed limit is set to inappropriate value, the force at the time of contact increases, and a serious personal injury could result. Risk assessment for the whole robot system is required to determine appropriate speed limit. Refer to followings about the risk according to the speed.

CR-35*i*A: MECHANICAL UNIT OPERATOR'S MANUAL(B-83734EN), Section 3.6 THE CHARACTERISTIC OF COLLABORATIVE ROBOT AND LIMITATIONS AND USAGE NOTES

CR-4*i*A, CR-7*i*A, CR-7*i*A/L: MECHANICAL UNIT OPERATOR'S MANUAL(B-83774EN), Section 3.6 THE CHARACTERISTIC OF COLLABORATIVE ROBOT AND LIMITATIONS AND USAGE NOTES

Use Payload Comp(configurable)

This item is displayed in the software version 7DC3/33 or later. Use when same payload acts on the robt every time during robot move. Refer to Section 7.5.

F2 [CONFIRM]

Pressing the F2 [CONFIRM] and the payload confirmation starts. The payload confirmation operation must be performed at least once after power on. The collaborative robot does not move until the payload confirmation is performed. Refer to Chapter 3.

- 1 If the collaborative robot settings are incorrectly, the safety function will not work, and a serious personal injury could result. When the collaborative robot settings are changed, the values must be verified and the function must be tested again.
- 2 When the contact stop is disabled, the collaborative robot does not stop even though the external force exceeds the limit, and a serious personal injury could result. If the robot system is designed to be able to disable the contact stop, adequate risk assessment for the whole robot system is necessary to verify that the contact stop can be disabled.
- 3 If "Password for CONFIRM" is set to DISABLE, anybody can perform the payload confirmation operation. If the payload confirmation operation is performed incorrectly, the external force is not detected correctly and safety function will not work, and a personal injury could result. When "Password for CONFIRM" is set to DISABLE, everybody who can operate the teach pendant must be authorized to perform the payload confirmation operation.

2.2.1 Verify Payload Setup

In the collaborative robot screen, select Payload setup: <DETAIL>, then the payload setup screen is displayed. In this screen, current parameters for the contact stop are displayed as followings. In this screen, you cannot change their parameters.

DCS								
	Payload setup							
			Status					
	PAYLOAD 1							
	PAYLOAD	[kg]		35.00	OK			
	CENTER X	[cm]		0.00	OK			
	CENTER Y	[cm]		0.00	OK			
	CENTER Z	[cm]		0.00	OK			
	INERTIA X	[kg	cm^2]	0.00	OK			
	INERTIA Y	[kg	cm^2]	0.00	OK			
	INERTIA Z	[kg	cm^2]	0.00	OK			
	PAYLOAD 2							
	PAYLOAD	[kg]		35.00	OK			
	[TYPE]							

PAYLOAD 1 to 9

Payload settings are displayed. In "SYSTEM > Motion" screen, you can change the payload settings.

DCS	5						
	Payload setup						
						Status	
	ARM LOAD	D #3					
	PAYLOA	D	[kg]		0.00	OK	
	CENTER	хя	[cm]		-3.50	OK	
	CENTER	łΥ	[cm]		0.00	OK	
	CENTER	εz	[cm]		8.00	OK	
	ARM LOAD	D #0					
	PAYLOA	D	[kg]		0.00	OK	
	CENTER	хя	[cm]		0.00	OK	
	CENTER	łΥ	[cm]		0.00	OK	
	CENTER	₹Z	[cm]		0.00	OK	
	ARM LOAD	D #0					
	[TYPE]						

ARM LOAD #axis_number

Arm load settings are displayed. In "SYSTEM > Motion" screen, you can change the payload settings.

DCS			
	Payload setup		
			Status
	Gravity vector		
	X [mm/s^2]	0.00	OK
	Y [mm/s^2]	0.00	OK
	Z [mm/s^2]	9800.00	OK
	Speed limit for push to	escape	
	J1 [deg/s]	9.00	OK
	J2 [deg/s]	9.00	OK
	Monitor interval(ms):	500	OK
	Vibration limit(N):	30.00	OK
	Payload error limit(N):	60.00	OK
	Status chk err lim(N):	60.00	OK
	Status chk time lim(s):	90.00	OK
	Vibration tolerance:	32	OK
	Payload change distance		
	Rotation limit (deg):	0.20	OK
	[TYPE]		

Gravity vector

Gravity vector is displayed. The gravity vector is changed by the mount angle of the robot.

Speed limit for push to escape

Speed limit in Push To Escape is displayed.

Monitor interval

Sampling time of the external force monitor is displayed. Refer to the section 4.2.

Vibration limit

The force limit by vibration is displayed.

Payload error limit

Payload error limit is displayed. When the error of the payload setting exceeds the payload error limit, the robot can not move. Refer to the subsection 4.2.2.

Status chk error lim

This is "Status check error limit", internal parameter for the auto status check.

Status chk time lim

This is "Status check time limit", the limit of time interval for the auto status check.

Vibration tolerance

Tolerance of floor vibration

Payload change distance Rotation limit (deg)

The orient rotation limit of the payload change distance. This item is displayed in the software version before 7DC3/29.

If the payload setting is incorrect, the safety function will not work correctly, and a personal injury could result. When payload settings are changed, the values must be verified and the function must be tested again.

2.3 OUTPUT SIGNAL OF COLLABORATIVE ROBOT STATUS

The state of the collaborative robot could be output as the digital output (DO, RO, Flag) by set the system variables. This is useful for programming and visual indicator.

SAFE state output

SETTING

\$DCSS_CLLB[1].\$DOTYP_SAFE: Set the type of output (DO:2, RO:9, F:35) \$DCSS_CLLB[1].\$DOIDX_SAFE: Set the index of output

SPECIFICATION

ON: The contact stop function is enabled and the safety is confirmed. In this case, the robot can move as usual.

STOP state output

SETTING

\$DCSS_CLLB[1].\$DOTYP_STOP: Set the type of output (DO:2, RO:9, F:35) \$DCSS_CLLB[1].\$DOIDX_STOP: Set the index of output

SPECIFICATION

ON: The contact stop function is enabled and the safety is not confirmed. In this case, the robot cannot move. The signal is ON in the following cases.

- The payload number is changed, but the payload change distance is disabled.
- The intensity of the external force is high.
- The auto status check has not been performed for Status check time limit. Refer to the subsection 4.2.1.

DSBL state output

SETTING

\$DCSS_CLLB[1].\$DOTYP_DSBL: Set the type of output (DO:2, RO:9, F:35) \$DCSS_CLLB[1].\$DOIDX_DSBL: Set the index of output

SPECIFICATION

ON: The contact stop function is disabled. This signal is useful for a visual indication when the robot is in collaborative operation.

ESCP state output

SETTING

\$DCSS_CLLB[1].\$DOTYP_ESCP: Set the type of output (DO:2, RO:9, F:35) \$DCSS_CLLB[1].\$DOIDX_ESCP: Set the index of output SPECIFICATION

ON: In Push To Escape.

Payload changing output

SETTING

\$DCSS_CLLB[1].\$DOTYP_PLCHG: Set the type of output (DO:2, RO:9, F:35) \$DCSS_CLLB[1].\$DOIDX_PLCHG: Set the index of output

SPECIFICATION

- ON: The contact stop function is disabled and the safety is not confirmed. In this case, if the contact stop is changed to enabling, the state would transition to STOP state, and the robot would stop. The signal is ON in the following cases.
 - The payload number is changed when the payload change distance is enabled.
 - The intensity of the external force is high.
 - The auto status check has not been performed for Status check time limit. Refer to the subsection 4.2.1.

Auto status check Time Out output

SETTING

\$DCSS_CLLB[1].\$DOTYP_TMOUT: Set the type of output (DO:2, RO:9, F:35) \$DCSS_CLLB[1].\$DOIDX_TMOUT: Set the index of output

SPECIFICATION

ON: The auto status check has not been performed for Status check time limit. Refer to the subsection 4.2.1.

Program Running output

SETTING

\$DCSS_CLLB[1].\$DOTYP_RUN: Set the type of output (DO:2, RO:9, F:35) \$DCSS_CLLB[1].\$DOIDX_RUN: Set the index of output

SPECIFICATION

ON: The program is running or the auto resume function is waiting to resume. This keeps ON when the program is resumed by Auto resume function. Refer to Section 7.1.

Program Paused output

SETTING

\$DCSS_CLLB[1].\$DOTYP_PAUSE: Set the type of output (DO:2, RO:9, F:35) \$DCSS_CLLB[1].\$DOIDX_PAUSE: Set the index of output

SPECIFICATION

ON: The program is paused and the auto resume function is not waiting to resume. Refer to Section 7.1.

Collaborative status output is not a Safe I/O. If you assign it to Safe I/O used by the safety function, the safety function will not work under some fault conditions and serious personal injury could result. Do not use the collaborative status output for safety purposes.

3

PAYLOAD CONFIRMATION OPERATION AFTER POWER ON

You must confirm that the payload setting is consistent with the actual payload at least once after power on the controller. This operation is called as Payload confirmation. The robot cannot move before Payload confirmation. In the collaborative robot screen, pressing F2[CONFIRM] and the payload confirmation process starts.

If the payload confirmation operation is performed incorrectly, the external force is not detected correctly and the safety function will not work, and a personal injury could result. When the payload confirmation operation is performed, the actual payload of the robot must be confirmed correctly, and anybody must not contact the robot.

Payload confirmation Operation

- If the message "Code number (master)" is displayed, enter the master code number for DCS. If the code number is not correct, the payload confirmation operation is failed.
 If the "Password for CONFIRM" is DISABLE in the collaborative robot screen, it is not necessary to enter code number.
- 2 First question "Actual payload is No. X?" is displayed.
 - Confirm the actual payload of the robot hand/tool/workpiece is surely equal to payload No. X.
 - If No. X is correct, press F4 [YES] and next question will be displayed.
 - Else, press F5 [NO] and the payload confirmation operation is failed. Change the payload setting and try again.
- 3 Second question "Nobody contacts with the robot?" is displayed. Confirm nobody contacts the robot, no foreign material is on the robot, and there is no external force to the actual robot.
 - If nobody contacts with the robot, press F4 [YES] and the result of the payload confirmation will be displayed.
 - Else, press F5 [NO] and the payload confirmation is failed. Remove the external force to the robot and try again.

The result of the payload confirmation

- "Payload confirmation success"
- Payload confirmation operation is completed. You can use the collaborative robot after this.

"Payload confirmation failed"

Remove the external force to the robot and try again. For example, a floor vibration may cause the external force.

If "Password for CONFIRM" is set to DISABLE, anybody can perform the payload confirmation operation. If the payload confirmation operation is performed incorrectly, the external force is not detected correctly and safety function will not work, and a personal injury could result. When "Password for CONFIRM" is set to DISABLE, everybody who can operate the teach pendant must be authorized to perform the payload confirmation operation.

3.1 Payload Confirmation without Teach Pendant

When you would like to perform the payload confirmation operation with some kind of terminal other than the teach pendant, configure "CONFIRM by DI" in the collaborative robot screen. This configuration is displayed in the software version 7DC3/30 or later.

DCS	3									
Collaborative robot										
CONFIRM by DI:			D	ISA	BLE	OK				
	CONFI	RM input:	D	I[0]					
	Payload No. output: GO[0]									
	[]	00177701		1						
	[TYPE]	CONFIRM	PEAKCLR			UNDO				

CONFIRM by DI (configurable)

When this item is set to ENABLE, you can perform the payload confirmation operation with an external signal. If it is ENABLE, you can also perform the payload confirmation operation with the teach pendant.

CONFIRM input (configurable)

When "CONFIRM by DI" is ENABLE and you change CONFIRM from OFF to ON, the payload confirmation will be performed. When the CONFIRM input changes from OFF to ON, you need to confirm that the actual payload of the robot hand/tool/workpiece is surely equal to current payload setting and nobody contacts with the robot. When CONFIRM input switches over and over in a short time, the payload confirmation may be performed incorrectly. Keep ON state more than 1 second.

Payload No. output (configurable)

Current payload setting number is output to a group output signal or a register set in this item. When you perform the payload confirmation operation with some kind of terminal other than the teach pendant, you can confirm the actual payload of the robot hand/tool/workpiece is surely equal to current payload setting by display of current payload setting number on the terminal.

If "CONFIRM by DI" is set to ENABLE, you set CONFIRM input to ON, it is necessary to confirm that the actual payload of the robot matches current payload setting and anybody does not contact the robot. Measure is necessary to prevent putting CONFIRM input by unauthorized person such as locking. If the payload confirmation operation is performed incorrectly, the external force is not detected correctly and safety function will not work, and a personal injury could be result.

4 SPECIFICATION OF COLLABORATIVE FUNCTION

This chapter introduces functions of the collaborative robot.

4.1 CONTACT STOP FUNCTION

When the external force exceeds the active force limit, the robot will stop.



After a contact stop, the robot cannot moves until the intensity of the external force becomes low.

When multiple programs are executed by multitasking function, all programs pause by the contact stop function. If you don't want to stop a program by the contact stop, set Interruption Disable in the program detail screen. Refer to OPERATOR'S MANUAL (Basic Operation) (B-83284EN), Chapter of PROGRAM STRUCTURE.

🕂 WARNING

Motion groups other than Collaborative robot are outside of the scope of the contact stop function. If a person comes into contact with the motion group other than Collaborative robot, a serious personal injury could result. If the robot system is designed to include the motion group other than Collaborative robot, adequate risk assessment for the whole robot system is necessary to verify that the motion group other than Collaborative robot are outside of the scope of the contact stop function.

4.1.1 Contact Stop in Teach Mode

The contact stop function is also enabled in T1/T2 mode. When the contact stop occurs, you cannot jog the robot in the usual manner until the intensity of the external force becomes low. This section provides the way to escape the robot by jogging in the contact stop situation. If the intensity of the external force does not become low, for example the robot contacts fixed object, you cannot jog in the usual manner. To recover this STOP situation, there are different manners between "ENABLE" and "ENBL(Shift+Reset)" set in the collaborative robot screen.

ENABLE

To recover from the STOP situation, you can jog the robot to the direction to go back to before contact. Hold down SHIFT key, press RESET key, and jog the robot without releasing SHIFT key. If the robot moves to the other direction, the robot stops immediately.

ENBL(Shift+Reset)

To recover from the STOP situation, hold down SHIFT key and press RESET key, then the contact stop function is disabled, and you can jog the robot. If the SHIFT key is released, or the intensity of the external force becomes low, the contact stop function is enabled again.

- In ENBL (Shift+Reset) case, when the contact stop is disabled by holding SHIFT key and pressing RESET key, even if an operator moves the robot to a dangerous direction by jogging, the robot will not stop.
- 2 When "ENBL (Shift+Reset)" is selected, the contact stop might be disabled when you hold down SHIFT key and press RESET key in T1 or T2 mode. If "ENBL (Shift+Reset)" is used, adequate risk assessment for the whole robot system is necessary to verify that the contact stop can be disabled when you hold down SHIFT key and press RESET key in T1 or T2 mode.

4.2 COLLABORATIVE ROBOT MONITOR

To verify that the contact stop function works correctly, the status of Collaborative robot is monitored periodically. If the status is possibly-abnormal, the robot stops until the external force data is confirmed normal status.

4.2.1 Auto Status Check

For the safety confirmation of Collaborative robot function, Auto Status Check is performed periodically.

- While the robot moving, the auto status check will not be performed. That is, when all motors of the robot stops, the auto status check is performed, even if the program is running.
- When the external force is large, the auto status check is not performed.
- When the elapsed time from a previous auto status check is longer than the reset time limit, the robot stops until the auto status check is performed. About Status check time limit, refer to the subsection 2.2.1.

Collaborative robot cannot move continuously more than Status check time limit. It is necessary to program that the robot stops in a state that does not receive any external force. Or you can use the auto resume function to avoid the program continues to be paused. Refer to Section 7.1.

4.2.2 Payload Monitor

If the payload setting is inconsistent with the actual payload, the sensor cannot detect the external force correctly. In this case, the robot stops until the intensity of the external force becomes low. For example, when the robot drops a workpiece unexpectedly, the robot cannot move.

When the robot picks or places a workpiece, an actual payload would be inconsistent with the payload setting temporarily. Then the sensor cannot also detect the external force correctly. In this case, the payload change distance is useful. Please refer to Chapter 5.

4.2.3 Collaborative Robot Monitor Screen

"Force Monitor" and "Payload Monitor" are displayed in the collaborative robot screen in the software version 7DC3/30 or later. You can monitor an external force and a load applied to the robot.

DCS	5					
	Collabo	rative ro				
	Force M	onitor:	Current	Peak		
		0.00[]	N](0%) 0%		
	Settin	gs :	Warning	Output		
			0%	GO[0]	
	Payload	Monitor:	Current	Peak		
	Force	:	0%	0%		
	Moment	t:	0%	0%		
	Setti	ngs :	Warning	Output		
	For	ce :	50%	GO[0]	
	Mon	ent :	509	k GO[0]	
	War	ning in D	DISA	ABLE		
		-			-	
	[TYPE]	CONFIRM	PEAKCLR		UNDO	

In the software version 7DC3/31 or later, direction components based on the robot world frame is displayed.

DCS								
	Collaborative robot							
	Force M	onitor:						
		Result	ant	х	Y	Z		
	Curr:	0.0	0[N]					
		()%	0%	0%	0%		
	Peak:	()%	0%	0%	0%		
	Output	t: GO[(0] GO[0] 0	3O[0] G	[0]C		
	Warniı	ng:	0%					
	Payload	Monitor	:					
	FORCE	Result	ant	х	Y	Z		
	Curr:	()%	0%	0%	0%		
	Peak:	()%	0%	0%	0%		
	Output	t: GO[0] GO[0]0	3O[0] G	0] 0]		
	Warniı	ng: 5	0%					
	MOMENT	Resul	ltant	х	Y	Z		
	Curr:	()%	0%	0%	0%		
	Peak:	()%	0%	0%	0%		
	Output	t: GO[0] GO[0]0	3O[0] G	0] 0]		
	Warniı	ng: 5	0%					
Warning in DSBL: DISABLE								
	[TYPE]	CONFIRM	I PEAK	CLR		UNDO		

In the software version 7DC3/33 or later, the compensation value of Variable Payload Compensation function is also displayed. Refer to Section 7.5.

CS									
	Collabo	rative ro	bot						
	Force M	onitor:							
		Resulta	nt	х	Y		z		
	Curr:	0.00	[N]						
		0%		0%	(0%	C)%	
	Peak:	0%		0%	()%	C)%	
	Output	t:							
	Curr	-: GO[0)] GO[0]	GO [0]	GO [0]	
	Warniı	ng: 0%	5						
	Payload	Monitor:							
	FORCE	Resulta	nt	х	Y		z		
	Curr:	0%		0%	()%	C)%	
	Comp:			0%	C	18	0	8	
	Peak:	0%		0%	()%	C)%	
	Output	t:							
	Curr	GO[0] GO	[0] GO [0]	GO[0]	
	Comp	?:	GO	[0]] GO[0]	GO [0]	
	Warniı	ng: 50%	8						
	MOMENT	Result	ant	х		Y	2	2	
	Curr:	0%		0%	()%	C)%	
	Comp:			0%	C	18	0	8	
	Peak:	0%		0%	()%	C)%	
	Output	t:							
	Curr	GO[0] GO	[0] GO[0]	GO[0]	
	Comp	? :	GO	[0]] GO[0]	GO[0]	
	Warniı	ng: 50%	8						
	Warning	g in DSBL	:		DIS	ABLE			
	[TYPE]	CONFIRM	PEAK	٦T.R			TIN	DO	1

Force Monitor

External force applied to the robot is displayed and output to external signals. You can monitor instant force such that a human contact.

Curr[N]

Current external force applied to the robot is displayed.

Curr[%]

Current force / Active Force Limit x 100 (When the contact stop function is disabled, the value is 0 %.) When the value is 100 %, a contact stop occurs. (SYST-320 "Contact force exceeds limit" is posted.)

Payload Monitor

Steady load applied to the robot is displayed and output to external signals. When the payload setting does not match the actual payload, this value will be large.

Force: Curr [%]

Averaged force / tolerance x 100 (During the payload change, the value is 0 %.)

When the value is 100 %, payload error occurs. (SYST-325 "Payload error is detected" is posted.) Force: Comp [%]

Force Compensation value during playback of Variable Payload Compensation function. Refer to Section 7.5.

Moment: Curr [%]

Averaged moment / tolerance x 100 (During the payload change, the value is 0 %.)

When the value is 100 %, payload error occurs. (SYST-325 "Payload error is detected" is posted.)

Moment: Comp [%]

Moment Compensation value during playback of Variable Payload Compensation function. Refer to Section 7.5.

Peak

Peak hold value of each item is displayed. Press F3[PEAK CLR] to clear the value.

4. SPECIFICATION OF COLLABORATIVE FUNCTION

Warning

When the current value[%] exceeds each warning value[%], following warning occurs.

- SYST-350 Force Monitor warning
- SYST-348 Payload Monitor (Force) warning
- SYST-349 Payload Monitor (Moment) warning

Set integer value from 0 to 100. When the value is 0, warning does not occur.

Warning in DSBL

DISABLE: No warning occurs when the contact stop function is disabled. ENABLE: Warnings also occur when the contact stop function is disabled.

Output

Each item can be output to signals etc. Following signal types can be set.

GO: Current value is output to Group output.

R: Current value is output to Numeric Register.

DO/RO: When the current value exceeds the set warning value, signal is set to ON.

4.3 RETREAT AFTER CONTACT STOP

When the robot was stopped by the contact stop and if strong force remained, the robot will retreat slightly.



Condition

Only at the contact stop in execution of the program (not in jogging)

How to disable the function

Set the system variable "\$DCSS_CLLB[1].\$REV_ENB" to FALSE.

4.4 PUSH TO ESCAPE

When an operator pushes the robot, the robot moves to pushed direction. Only J1 and J2 can respectively escape by pushing. In Push To Escape, TCP orientation is not kept.



Condition

- AUTO mode
- Any alarm is not posted.
- The robot stops.
- The contact stop function is enabled.
- The external force is less than the force limit for Push To Escape.

How to disable the function

Set the system variable "\$DCSS_CLLB[1].\$PTE_ENABLE" to FALSE.

Tips for pushing

- When you push the robot too strongly, the robot cannot escape.
- Farther and farther from the rotation axis, it is easy to push.
- If the robot does not escape, release your hand for a moment and try again.

5 PROGRAMMING

You can teach a program of the collaborative robot in much the same way as other FANUC robots except for situations such that the collaborative robot is subjected to external force.

For example, while the robot picks or places a workpiece, the robot is subjected to external force. The payload number is changed by PAYLOAD instruction in program. The timing of PAYLOAD instruction execution is not the same as the timing of the actual workpiece picking/placing. Therefore a contact stop may occur by mismatch between the payload setting and the actual payload. To avoid the unnecessary contact stop during the payload change, the contact stop function can be disabled.

In this situation, to prevent a program from being paused by the contact stop, you should set the contact stop function to disable in limited area. There are following ways to disable the contact stop function.

- To set all disabling I/O of force limits to ON
- To change payload number when the payload change distance enabling input

The disabling I/O of force limits and the payload change distance enabling could be assigned with the safety I/O (e.g. Cartesian position check, Joint speed check, etc.). The safety I/O is helpful to teach a safety program.

- 1 When the contact stop is disabled, the collaborative robot does not stop even though the external force exceeds the limit, and serious personal injury could result. If the robot system is designed to be able to disable the contact stop, adequate risk assessment for the whole robot system is necessary to verify that the contact stop can be disabled.
- 2 Opening and closing of a robot hand could do a human serious personal injury. When you design a robot hand, adequate risk assessment for the robot hand is necessary to verify that the robot hand do a human no injury.

This chapter provides some examples of programming the collaborative robot.

5.1 PICKING WORKPIECE

5.1.1 Simple Picking Program

This section provides a setting to pick a workpiece. In following way, the contact stop is disabled just after payload number change. If the robot moves more than the payload change distance, the contact stop is enabled.

When the contact stop is disabled, the collaborative robot does not stop even though the external force exceeds the limit, and serious personal injury could result. If the robot system is designed to be able to disable the contact stop, adequate risk assessment for the whole robot system is necessary to verify that the contact stop can be disabled.

- 1 Press the [MENU] key.
- 2 Select SYSTEM.
- 3 Press F1, [TYPE].
- 4 Select DCS.
- 5 If neither DCS top screen nor Collaborative robot screen, press [PREV] key, and select Collaborative robot.
- 6 Scroll down. You will see a screen similar to the following.

DCS	5								
	Collaborative robot								
	Limit	1: 105	.00	-[0]	OK				
	Limit	2: 0	.00	·[0]	OK				
	Limit	3: 0	.00	·[0]	OK				
	Limit	4: 0	.00	·[0]	OK				
	Escape	∍: 300	.00		OK				
	Payload	change d	istance (mm)					
		Current	+Limit	-Limit					
	x:	0.00	0.00	0.00	OK				
	Y:	0.00	0.00	0.00	OK				
	z:	0.00	0.00	0.00	OK				
	Enabl	ing input	: ON	[0]	OK				
	[TYPE]	CONFIRM			UNDO				

7 Set Payload change distance +Limit and –Limit of each direction(X, Y, Z). In this example, +Limit of Z = 3.0mm, and the limit of other directions is 1.0mm.

DCS	3								
	Collaborative robot								
	Limit	1: 105	.00	-[0]	OK				
	Limit	2: 0	.00	-[0]	OK				
	Limit	3: 0	.00	-[0]	OK				
	Limit	4: 0	.00	-[0]	OK				
	Escape	e: 300	.00		OK				
	Payload	change d	istance ((mm)					
		Current	+Limit	-Limit					
	X:	0.00	1.00	1.00	CHGD				
	Y:	0.00	1.00	1.00	CHGD				
	z:	0.00	3.00	1.00	CHGD				
	Enabl	ing input	: ON	[0]	OK				
	[TYPE]	CONFIRM			UNDO				

8 Apply to DCS parameter. Refer to Dual Check Safety Function OPERATOR'S MANUAL (B-83184EN), Chapter of OVERVIEW, Section of APPLY TO DCS PARAMETER.

In above setting, when the payload number is changed, the contact stop is disabled. When the flange position of the robot moves to +Z direction of the world coordinate system more than 3mm or other direction more than 1mm from the position the payload number is changed, the contact stop returns to enabled.



If the contact stop returns to enabled before the external force is sufficiently small, the program will be paused. Adjust the setting as followings, and prevent the program from being paused.

- Reduce the speed of the movement the robot lift up a workpiece.
- Wait a little after lifting up a workpiece.
- Set the payload change distance to sufficient distance to the external force is small.

The payload change distance is 0 by default. If you change the payload change distance, adequate risk assessment for the whole robot system is necessary to verify that the payload change distance does not cause any dangerous situation.

- 1 When the contact stop is disabled, the collaborative robot does not stop even though the external force exceeds the limit, and serious personal injury could result. If the robot system is designed to be able to disable the contact stop, adequate risk assessment for the whole robot system is necessary to verify that the contact stop can be disabled.
- 2 In this setting, when the payload number is changed, the contact stop function is disabled anywhere in the envelope of the robot. This indicates the contact stop function is disabled unexpectedly, if any mistakes.

5.1.2 Limiting Picking Area

In previous sample program, when the payload number is changed, the contact stop function is disabled anywhere in the envelope of the robot. This indicates the contact stop function is disabled unexpectedly, if any mistakes. For example, the payload number is changed by a mistake of programming.

This section provides additional setting to specify the area to payload change. In this setting, even if the payload number is changed out of the zone, the contact stop function still is enabled. You can program as below.

- Set the DCS tool model for the payload change.
- Set the zone for the payload change to the DCS Cartesian Position Check.
- Set Safe I/O connect
- Set Safe I/O of the zone as the payload change distance enabling input.



SOFTWARE OPTION

The following options are necessary for below setting.

- J567 DCS Position/Speed check function
- J568 DCS Safe I/O connection function

Refer to Dual Check Safety Function OPERATOR'S MANUAL (B-83184EN) for more detail information about these options. Below setting is one example.
SETTING

DCS Tool model setting

- 1 Press [MENU] key.
- 2 Select SYSTEM.
- 3 Press F1, [TYPE].
- 4 Select DCS.
- 5 If neither DCS top screen nor User model, press [PREV] key several times, and select User model.
- 6 Select No.1 and press F3, [DETAIL]. You will see a screen similar to the following.

DCS	5							
	User el	ement	detail	L				
	No. 1	- [***	*****	*****	*****	****	***]	
	Element	: 1	Sta	tus:	OK			
	1 Ena	able/D	isable	:			DISABLE	
	2 Lir	nk No.	(99:F	acePla	ate) :		99	
	3 Lir	nk typ	e:				NORMAL	
	4 Too	ol fra	me:				0	
	5 Sha	ape:					Point	
	6 Si2	ze (mm):				0.0	
	Pos1							
	7 X	: (0.000	Y:	0.000	z:	0.000	
	Pos2							
	8 X	: (0.000	Y:	0.000	z:	0.000	
	[TYPE]						UNDO	>

7 Set the TCP model as below. Please set the Tool frame position in (X, Y, Z) of Pos1. In this example, size is 50mm.

DCS		
	User element detail	
	No. 1 [***********************************	*****]
	Element: 1 Status: CHGD	
	1 Enable/Disable:	ENABLE
	2 Link No. (99:FacePlate) :	99
	3 Link type:	NORMAL
	4 Tool frame:	0
	5 Shape:	Point
	6 Size (mm):	50.0
	Pos1	
	7 X: 0.000 X: 0.000 Z	• 0.000
	Pog2	
	8 X. 0 000 X. 0 000 7	• 0 000
	5 A. 0.000 I. 0.000 Z	
	[TYPE]	UNDO >

8 Apply to DCS parameter. Refer to Dual Check Safety Function OPERATOR'S MANUAL (B-83184EN), Chapter of OVERVIEW, Section of APPLY TO DCS PARAMETER.

DCS Cartesian Position Check setting

- 1 Press [MENU] key.
- 2 Select SYSTEM.
- 3 Press F1, [TYPE].
- 4 Select DCS.
- 5 If neither DCS top screen nor Cart. position check, press [PREV] key, and select Cart. position check.
- 6 Select No.1 and press F3, [DETAIL]. You will see a screen similar to the following.

DCS	;									
	Cartesi	an positi	on check							
	No.	1	S	tatus:						
	1 Co	mment: [*******	*******	********]					
	2 Ena	able/Disab	ole:	DIS	ABLE					
	3 Met	3 Method(Safe side):Diagonal(IN)								
	4 Gro	oup:		1						
	5 Ta:	rget model	1:		-1					
	6 Ta:	rget model	2:		0					
	7 Ta:	rget model	3:		0					
		(0:Disable	e, -1:Rob	ot, -2:To	ol)					
	8 Us	er frame:			0					
	Pos	sition(mm)	:							
		Current	Poi	nt 1	Point 2					
	9 X	1110.0		0.0	0.0					
	10 Y	0.0	(0.0	0.0					
	11 Z	940.0		0.0	0.0					
	12 St	op type:		Power-Of	f stop					
	13 di	sabling in	nput:	-[0:]				
				-						
	[TYPE]	PREV	NEXT	RECORD	UNDO	>				

- 7 Set the zone for the payload change such as below.
 - Set target model to TCP model.
 - Set Stop type to Not stop. The robot can move both inside and outside this zone.
 - Set the Position for the payload change. Please note that the contact stop function is disabled when TCP is in this zone.

DCS										
	Cartesi	an positi	on check							
	No	. 1	S	tatus:CHG	\$D					
	1 Co:	mment: [*******	*******	*****]					
	2 En	able/Disab	ole:	ENAI	BLE					
	<pre>3 Method(Safe side):Diagonal(IN)</pre>									
	4 Gr	1								
	5 Ta:	rget model	. 1:		1					
	6 Ta	rget model	2:		0					
	7 Ta:	rget model	3:		0					
		(0:Disable	e, -1:Rob	ot, -2:To	ol)					
	8 Us	er frame:			0					
	Pos	sition(mm)	:							
		Current	Poi	nt 1	Point 2					
	9 X	1110.0	100	0.0	800.0					
	10 Y	0.0	-10	0.0	100.0					
	11 Z	940.0	60	0.0	800.0					
	12 St	op type:		Not stop						
	13 di	sabling ir	nput:	-[0:]				
	[TYPE]	PREV	NEXT	RECORD	UNDO	>				

8. Apply to DCS parameter. Refer to Dual Check Safety Function OPERATOR'S MANUAL (B-83184EN), Chapter of OVERVIEW, Section of APPLY TO DCS PARAMETER.

Safe I/O connect setting

- 1 Press [MENU] key.
- 2 Select SYSTEM.
- 3 Press F1, [TYPE].
- 4 Select DCS.
- 5 If neither DCS top screen nor Safe I/O connect, press [PREV] key several times, and select Safe I/O connect. You will see a screen similar to the following.

DCS								
	Safe I/0) con	nect					
	Output	Inpu	Input2 Status					
	[0]=		[0]	[0]	OK	
	[0]=		·[0]	[0]	OK	
	[0]=		·[0]	[0]	OK	
	[0]=		[0]	[0]	OK	
	[0]=		[0]	[0]	OK	
	[0]=		·[0]	[0]	OK	
	[0]=		[0]	[0]	OK	
	[0]=		[0]	[0]	OK	
	[0]=		[0]	[0]	OK	
	[0]=		[0]	[0]	OK	
	[TYPE]			CLEAR		UN	DO	

6 Set SIR[1] = CPC[1] (Cartesian Position Check of the zone for the payload change).

DCS	5							
	Safe I/0) con	nect					
	Output		Input1	L	Input2 Status			
	SIR[1]=	CPC[1]	[0] CHGD)	
	[0]=	[0]	[0] OK		
	[0]=	[0]	[0] OK		
	[0]=	[0]	[0] OK		
	[0]=	[0]	[0] OK		
	[0]=	[0]	[0] OK		
	[0]=	[0]	[0] OK		
	[0]=	[0]	[0] OK		
	[0]=	[0]	[0] OK		
	[0]=	[0]	[0] OK		
	[TYPE]		(CLEAR		UNDO		

7 Apply DCS parameter. Refer to Dual Check Safety Function OPERATOR'S MANUAL (B-83184EN), Chapter of OVERVIEW, Section of APPLY TO DCS PARAMETER.

When you would like to set plural zones for the payload change, you can set such as followings.

DCS									
	Safe I/0) con	nect						
	Output	Inpu	ıt1		Inpui	Input2 Status			
	SIR[2]=	CPC	2[2]	OR	CPC[3]	CHGD	
	SIR[1]=	CPC	2[1]	OR	SIR[2]	CHGD	
	[0]=		-[0]		[0]	OK	
	[0]=		-[0]		[0]	OK	
	[0]=		-[0]		[0]	OK	
	[0]=		-[0]		[0]	OK	
	[0]=		-[0]		[0]	OK	
	[0]=		-[0]		[0]	OK	
	[0]=		-[0]		[0]	OK	
	[0]=		-[0]		[0]	OK	
	[TYPE]			CLEA	R		UN	IDO	

Payload change distance Enabling input

- 1 Press [MENU] key.
- 2 Select SYSTEM.
- 3 Press F1, [TYPE].
- 4 Select DCS.
- 5 If neither DCS top screen nor Collaborative robot screen, press [PREV] key, and select Collaborative robot.
- 6 Scroll down. You will see a screen similar to the following.

DCS	3					
	Collabo	rative ro	bot			
	Limit	1: 105	.00	-[0]	OK	
	Limit	2: 0	.00	-[0]	OK	
	Limit	3: 0	.00	-[0]	OK	
	Limit	4: 0	.00	-[0]	OK	
	Escape	e: 300	.00		OK	
	Payload	change d	istance	(mm)		
		Current	+Limit	-Limit		
	X:	0.00	1.00	1.00	OK	
	Y:	0.00	1.00	1.00	OK	
	Z:	0.00	3.00	1.00	OK	
	Enabl:	ing input:	: ON	[0]	OK	
	[TYPE]	CONFIRM			UNDO	

7 Set Payload change distance Enabling input to SIR[1].

DCS	5					
	Collabo	rative ro	bot			
	Limit	1: 105	.00	-[0]	OK	
	Limit	2: 0	.00	[0]	OK	
	Limit	3: 0	.00	[0]	OK	
	Limit	4: 0	.00	[0]	OK	
	Escape	∋: 300	.00		OK	
	Payload	change d	istance (mm)		
		Current	+Limit	-Limit		
	x:	0.00	1.00	1.00	OK	
	Y:	0.00	1.00	1.00	OK	
	z:	0.00	3.00	1.00	OK	
	Enabl	ing input	: SIR	[1]	CHGD	
	[TYPE]	CONFIRM			UNDO	

8 Apply to DCS parameter. Refer to Dual Check Safety Function OPERATOR'S MANUAL (B-83184EN), Chapter of OVERVIEW, Section of APPLY TO DCS PARAMETER.

In above setting, when the robot is out of the safety area, the contact stop is not disabled by the payload number change. To avoid the contact stop disabled in unexpected area, set the area as above.

5.2 DISABLING CONTACT STOP IN PROGRAM

This section provides how to disable the contact stop function by program. This is useful in a system such that the robot should stay in movement even though the robot is subjected to external force in specified area.

When the contact stop is disabled, the collaborative robot does not stop even though the external force exceeds the limit, and serious personal injury could result. If the robot system is designed to be able to disable the contact stop, adequate risk assessment for the whole robot system is necessary to verify that the contact stop can be disabled.

SOFTWARE OPTION

The following options are necessary for below setting.

- J567 DCS Position/Speed check function
- J568 DCS Safe I/O connection function

Refer to Dual Check Safety Function OPERATOR'S MANUAL (B-83184EN) for more detail information about these options. Below setting is one example.

SETTING

DCS Tool model setting

- 1 Press [MENU] key.
- 2 Select SYSTEM.
- 3 Press F1, [TYPE].
- 4 Select DCS.
- 5 If neither DCS top screen nor User model, press [PREV] key several times, and select User model.
- 6 Select No.1 and press F3, [DETAIL]. You will see a screen similar to the following.

DCS										
	User	elem	ent o	leta	ail					
	No.	1	[***	* * * *	******	****	* * *	* * * *	***]	
	Eleme	ent:	1	S	tatus:	ок				
	1	Enab	le/Di	sab	le:				DISABLE	
	2	Link	No.	(99	:FacePl	ate)	:		99	
	3	Link	type	:					NORMAL	
	4	Tool	fram	e:					0	
	5	Shap	e:						Point	
	6	Size	(mm)	:					0.0	
	Ро	s1								
	7	x:	0	.00	0 Y:	0.0	00	z:	0.000	
	Po	s2								
	8	x:	0	.00	0 Y:	0.0	00	z:	0.000	
	[TYPE	:]			DETAIL]	ROB	ОТ	UNDO	>

7 Set the TCP model as below. Please set the Tool frame position in (X, Y, Z) of Pos1. In this example, size is 50mm.

DCS							
	User el	Lement det	ail				
	No.	1 [*****	******	*******	****]		
	Element	:: 1	Status:	CHGD			
	1 En	able/Disa	ble:		ENABLE		
	2 Li	nk No. (9	9:FacePl	ate) :	99		
	3 Li	nk type:			NORMAL		
	4 To	ol frame:			0		
	5 Sh	ape:			Point		
	6 Si	ze (mm):			50.0		
	Pos1						
	73	K: 0.0	00 Y:	0.000 Z:	0.000		
	Pos2						
	8 X	K: 0.0	00 Y:	0.000 Z:	0.000		
	[TYPE]				UNDO	>	

8 Apply to DCS parameter. Refer to Dual Check Safety Function OPERATOR'S MANUAL (B-83184EN), Chapter of OVERVIEW, Section of APPLY TO DCS PARAMETER.

DCS Cartesian Position Check setting

- 1 Press [MENU] key.
- 2 Select SYSTEM.
- 3 Press F1, [TYPE].
- 4 Select DCS.
- 5 If neither DCS top screen nor Cart. position check, press [PREV] key, and select Cart. position check.
- 6 Select No.1 and press F3, [DETAIL]. You will see a screen similar to the following.

DCS	5						
	Cart	esia	n positio	on check			
		No.	1	S	tatus:		
	1	Com	ment: [*******	*******	******]	
	2	Enak	ole/Disab	le:	DIS	ABLE	
	3	Meth	nod(Safe	side):Dia	agonal(IN)	
	4	Grou	ıp:			1	
	5	Targ	get model	1:		-1	
	6	Targ	get model	2:		0	
	7	Targ	get model	3:		0	
		((Disable	, -1:Rob	ot, -2:To	ol)	
	8	User	frame:			0	
		Posi	tion(mm)	:			
		C	Current	Poi	nt 1	Point 2	
	9	х	1110.0		0.0	0.0	
	10	Y	0.0	(0.0	0.0	
	11	z	940.0		0.0	0.0	
	12	Stor	p type:		Power-Of:	f stop	
	13	disa	abling in	nput:	-[0:]
			-			n	
	[TYPE	Ξ]	PREV	NEXT	RECORD	UNDO	>

- 7 Set the zone for the contact stop disable
 - Set target model to TCP model.
 - Set Stop type to Not stop. The robot can move both inside and outside this zone.
 - Set the Position for the contact stop disable.

DCS						
	Cartesi	an positi	on check			
	No.	1	S	tatus:CHG	¹ D	
	1 Cor	ment: [*******	*******	*****]	
	2 Ena	able/Disab	le:	ENA	BLE	
	3 Met	hod(Safe	side):Dia	agonal(IN)	
	4 Gro	oup:			1	
	5 Tai	get model	1:		1	
	6 Tai	get model	2:		0	
	7 Tai	get model	3:		0	
		(0:Disable	e, -1:Robe	ot, -2:To	ol)	
	8 Use	er frame:			0	
	Pos	ition(mm)	:			
		Current	Poi	nt 1	Point 2	
	9 X	1110.0	100	0.0	800.0	
	10 Y	0.0	-10	0.0	100.0	
	11 Z	940.0	60	0.0	800.0	
	12 Sto	op type:		Not stop		
	13 di:	sabling ir	nput:	-[0:]
	[TYPE]	PREV	NEXT	RECORD	UNDO	>

8 Apply to DCS parameter. Refer to Dual Check Safety Function OPERATOR'S MANUAL (B-83184EN), Chapter of OVERVIEW, Section of APPLY TO DCS PARAMETER.

DCS NSI setting

- 1 Press [MENU] key.
- 2 Select I/O.
- 3 Press F1, [TYPE].
- 4 Select Digital.
- 5 Press F2, [CONFIG], and press F3, [IN/OUT]. You will see a screen similar to the following.



6 Set unassigned DO to NSI. In this example, DO[70] = NSI[1]. NSI is RACK 36, SLOT 0.

I/C) Digi	tal (Dut						
	#		RANGE		RACK	SLOT	START	STAT.	
	1	DO[1-	64]	89	1	1	ACTIV	
	2	DO[65-	69]	0	0	0	UNASG	
	3	DO[70-	70]	36	0	1	PEND	
	4	DO[71-	512]	0	0	0	UNASG	
	[- 1				DIT			
1	[TYPI	S]]	MONITOR	IN	/OUT	DELI	CLE CLE	HELP	>

Safe I/O connect setting

- 1 Press [MENU] key.
- 2 Select SYSTEM.
- 3 Press F1, [TYPE].
- 4 Select DCS.
- 5 If neither DCS top screen nor Safe I/O connect, press [PREV] key several times, and select Safe I/O connect. You will see a screen similar to the following.

DCS							
	Safe I/	0 com	nect				
	Output		Input1		Inpu	t2 Stat	us
	SIR[1]=	[0]	[0] 0	к
	SIR[2]=	[0]	E	0]0	к
	SIR[3]=	[0]	[0] 0	к
	SIR[4]=	[0]	[0] 0	к
	SIR[5]=	[0]	[0] 0	к
	SIR[6]=	[0]	[0] 0	к
	SIR[7]=	[0]	[0] 0	к
	SIR[8]=	[0]	[0] 0	к
	SIR[9]=	[0]	[0] 0	к
	SIR[10]=	[0]	[0] C	ĸ
	[TYPE]		(CLEAR		UNDO	

6 Set SIR[1] = NSI[1] AND CPC[1] (Cartesian Position Check of the zone for the contact stop disable).

DCS	5								
	Safe I/	0 con	nect						
	Output		Input1			Inpu	t2 St	atus	
	SIR[1]=	NSI[1]	ANI	CPC	[1]	CHGI)
	SIR[2]=	[0]		[0]	OK	
	SIR[3]=	[0]		[0]	OK	
	SIR[4]=	[0]		[0]	OK	
	SIR[5]=	[0]		[0]	OK	
	SIR[6]=	[0]		[0]	OK	
	SIR[7]=	[0]		[0]	OK	
	SIR[8]=	[0]		[0]	OK	
	SIR[9]=	[0]		[0]	OK	
	SIR[10]=	[0]		[0]	OK	
		1							
	[TYPE]		C	CLEA	R		UN	DO	

7 Apply DCS parameter. Refer to Dual Check Safety Function OPERATOR'S MANUAL (B-83184EN), Chapter of OVERVIEW, Section of APPLY TO DCS PARAMETER.

Force limit disabling input setting

- 1 Press [MENU] key.
- 2 Select SYSTEM.
- 3 Press F1, [TYPE].
- 4 Select DCS.
- 5 If neither DCS top screen nor Collaborative robot screen, press [PREV] key, and select Collaborative robot. You will see a screen similar to the following.

DCS	5									
	Collabo	rative	roł	oot						
									status	
	Contact	stop	stat	us:			SAFE			
	Enable/	Disabl	e: I	INABLE	3				OK	
	Force s	ensor								
	Serial	L numbe	er:			2	xxxxx	xx	OK	
	Group:							1	OK	
	Payload	setup	:			<	DETAI	L>	OK	
	Active	Payloa	d: 1	No. 1	[**	**	*****	***	****]	
	Externa	l forc	e (1	1) /	Dis	ab	ling	inpu	ıt	
	Limit	1:	105	.00		Ē	0]		OK	
	Limit	2:	100	.00		Ē	0]		OK	
	Limit	3:	0.	00		[0]		OK	
	Limit	4:	Ο.	00		[0]		OK	
	Escape	••	300.	.00					OK	
	[TYPE]	CONFI	RM						UNDO	

6 Set the all set limits of disabling input to SIR[1].

DCS	}									
	Collabo	Collaborative robot								
							Status			
	Contact	stop sta	tus:			SAFE				
	Enable/I	Disable:	ENABLE	1			OK			
	Force se	ensor								
	Serial	number:			XX	xxxxxx	K OK			
	Group:					:	1 OK			
	Payload			<1	DETAIL	> 0K				
	Active H	Payload:	No. 1	[***	***	*****	******]			
	External	l force (N) /	Disa	abl	ing ir	nput			
	LIMIT	1: 105	.00	SIR	[1]	CHGD			
	LIMIT	2: 100	.00	SIR	[1]	CHGD			
	LIMIT	3: 0	.00	[[(0]	OK			
	LIMIT	4: 0	.00	[[(0]	OK			
	Escape	e: 300	.00				OK			
	[TYPE]	CONFIRM					UNDO			

7 Apply DCS parameter. Refer to Dual Check Safety Function OPERATOR'S MANUAL (B-83184EN), Chapter of OVERVIEW, Section of APPLY TO DCS PARAMETER.

SAMPLE PROGRAM

1: L P[1] 200mm/s CNT100
2: L P[2] 200mm/s CNT100
3: DO[70] = ON
4: L P[3] 50mm/s CNT100
5: L P[4] 50mm/s CNT100
6: DO[70] = OFF
7: L P[5] 200mm/s CNT100

In the above settings, the contact stop function is disabled when the robot is in the zone for the contact stop disable and DO[70] = ON. Therefore, in line 4 of the sample program, the contact stop function is disabled just after the robot enters into the zone. When the robot leaves the zone after that, the contact stop function is enabled.

If the external force is large and the contact stop function is enabled, the program will be paused. To avoid the situation, set the zone for the contact stop disable appropriately.



6

PERIODICAL CHECK

To check the force sensor accuracy, the periodical function check operation should be performed in proper interval. FANUC recommends once a month.

HOW TO CHECK

- Execute checking program according to the following.
- The robot program can be executed with the robot hand.

How to make the checking program

- 1 Make a program that the robot moves to some positions.
 - Term type of motion instruction is FINE.
 - Each postures of the robot in the checking program must have a distance from each other. This is for checking the output of all direction of the force sensor.
 - The maximum number of the position in the checking program is 30.
 - All operating time of the program must be less than 90 sec.
- 2 Insert CALL instruction to call FORCETST(parameter). FORCETST is KAREL program loaded by default.
 - In the first position, call FORCETST(1).
 - In the second position onward, call FORCETST(2).
 - In the last position, call FORCETST(3).

SAMPLE PROGRAM

The following is the example of the proper position data for the periodical function check.

CR-35*i*A



CR-4iA, CR-7iA, CR-7iA/L



The following is an example checking program.



RESULT OF CHECK

- If the execution of the checking program is completed without any alarms, the force sensor accuracy is good.
- If "SYST-325 Payload error is detected" occurs, check the payload setting at first. If the payload setting is correct, the force sensor may have problems.
- If "SYST-327 Inappropriate regular test program" occurs, the checking program has a problem. Please refer to above explanation.
 - Verify that each postures of the robot in the checking program must have a distance from each other.
 - Verify that the number of the positions of the checking program is less than 30.

7. UTILITY

This chapter explains special functions of the collaborative robot.

7.1 AUTO RESUME FUNCTION

Auto resume function automatically resumes the robot program paused by the contact stop function.

Condition

- In AUTO mode
- Teach Pendant is OFF.
- The program is paused by the contact stop function.
- Any alarm is not posted.

In following cases, the auto resume is canceled.

- The contact stop occurs continuously.
- An operator pushes the robot for a certain time.
- It has a certain time after the contact stop.
- The program was aborted.
- Push To Escape is performed.
- Retreat after the contact stop is performed.
- The paused program's group mask is not same as collaborative robot group.
- HOLD key was pressed after contact stop.

- 1 The auto resume function is disabled by default. If you enable this function, adequate risk assessment for the whole robot system is necessary to verify that the auto resume function does not cause any dangerous situation.
- 2 If the auto resume function is enable, the situation that the robot program is possible to be resumed automatically should be indicated clearly that the operator can notice from anywhere in the operation area. The program running output can be used to the indication; it is ON when the

robot program is possible to be resumed automatically. Refer to Section 2.3.

Setting up

You can set up the auto resume function with below system variables.

\$DCSS_CLLB[1].\$AUTO_RESUME

If it is set to 1, the auto resume function is enabled. If it is set to 0, the auto resume function is disabled.

Limitation

- Multi task program is not supported, because the auto resume can resume only one task. When multiple programs is executed by multitasking function, all programs pause by the contact stop function. If you don't want to stop a program by the contact stop, set Interruption Disable in the program detail screen. Refer to OPERATOR'S MANUAL (Basic Operation) (B-83284EN), Chapter of PROGRAM STRUCTURE.
- Auto resume function will not resume when the program group mask is not same as collaborative robot group setting. Therefore, CALL instruction may cause unintended resuming. For example, if using following two programs, the program will not be resumed when the cursor is in the PRG_A.TP but the program will be resumed when the cursor is in the PRG_B.TP. In this case, set Group mask of collaborative robot group to 1 in PRG_A detail screen, then the auto resume function will resumes PRG_A.

PRG_A.TP(Group Mask *,*,...)

1:CALL PRG_B
2:DO[1] = ON
3:WAIT DI[1] = ON
4:CALL PRG_B

PRG_B.TP(Group Mask 1,*,...)

1:J P[1] 10% CNT 100	
2:J P[2] 10% CNT 10	

7.2 HIGH SPEED MODE

By default, the speed of the collaborative robot is limited by limited speed value (CR-35iA: 250mm/s, CR-4iA, CR-7iA, CR-7iA/L: 750mm/s)

To use the collaborative robot as High Speed Mode, set the system variable \$CF_LIM_GRP[group number of collaborative robot].\$SPEEDLIM to the value less than the maximum speed of the robot . Then the speed of the collaborative robot is limited by the set speed. About the maximum speed of the robot, refer to the manual of each robot.

When the robot is CR-35*i*A and the software version is 7DC3/23 or later, you can switch on and off of the speed limit without override change. Use in accordance with the following configuration.

SAMPLE CONFIGURATION (CR-35*i*A before 7DC3/22, CR-4*i*A, CR-7*i*A, CR-7*i*A/L)

When the contact stop function is enable, TCP speed is checked regardless of the value of \$CF_LIM_GRP[group number of collaborative robot].\$SPEEDLIM. If the TCP speed exceeds the limit, the robot stops. For example, on CR-35*i*A, to avoid the stop, please set the override to a integer value less than "25000/\$CF_LIM_GRP[group number of collaborative robot].\$SPEEDLIM" when the contact stop function is enabled. For example, when \$CF_LIM_GRP[group number of collaborative robot].\$SPEEDLIM" stop robot].\$SPEEDLIM =750, set the override less than 33% (25000/750=33.333).

The following is the example of the robot system with high speed mode with the area sensor.

- When a human is in the Zone 1, the contact stop is enabled.
- When a human is in the Zone 2 (including Zone 1), the override is set to 33 % by Override Select function.
- When a human is not in Zone 2 (including Zone 1), the override is set to 100 % by Override Select function.



- Area sensor has 1 safety output and 1 non-safety output.
- The safety output of the area sensor is connected to SFDI1 of the robot controller. This signal is accessed as SPI[1] in DCS function. When a human is in Zone 1, the SPI[1] becomes OFF.
- The non-safety output of the area sensor is connected to digital input signal that is assigned to DI[20]. When a human is in the Zone 2, DI[20] becomes OFF.
- Set the disabling input of the force limit to SPI[1]. Refer to Section 2.2.

@Limit	1:	105.00	SPI[1]	OK
Limit	2:	0.00	[0]	OK
Limit	3:	0.00	[0]	OK
Limit	4:	0.00	[0]	OK

• Set the Override select as below. By this setting, the override is set to 100% when DI[20] is ON, and the override is set to 33% when DI[20] is OFF. (DI[21] is not used.)

OVERRIDE	SELECT			
1 F1	unction E	nable :ENAB	LE	
2 S:	ignal 1	: DI[20][OFF]	
3 S:	ignal 2	: DI[21][OFF]	
Si	gnal 1	Signal 2	Override	
4	OFF	OFF	33 %	
5	OFF	ON	33 %	
6	ON	OFF	100 %	
7	ON	ON	100 %	
[TYPE]			

When the contact stop is disabled, the collaborative robot does not stop even though the external force exceeds the limit, and a serious personal injury could result. If the robot system is designed to be able to disable the contact stop, adequate risk assessment for the whole robot system is necessary to verify that the contact stop can be disabled.

SAMPLE CONFIGURATION (CR-35*i*A after 7DC3/23)

The following is the example of the robot system with high speed mode with the area sensor.

- When a human is in the Zone 1, the contact stop is enabled.
- When a human is in the Zone 2 (including Zone 1), the speed limit is ON by turning OFF the disabling input of the speed limit.
- When a human is not in Zone 2 (including Zone 1), the speed limit is OFF by turning ON the disabling input of the speed limit.



- Area sensor has 1 safety output and 1 non-safety output.
- The safety output of the area sensor is connected to SFDI1 of the robot controller. This signal is accessed as SPI[1] in DCS function. When a human is in Zone 1, the SPI[1] becomes OFF.
- The non-safety output of the area sensor is connected to digital input signal that is assigned to DI[20]. When a human is in the Zone 2, DI[20] becomes OFF.
- Set the disabling input of the force limit to SPI[1]. Refer to Section 2.2.

@Limit	1:	105.00	SPI[1]	OK
Limit	2:	0.00	[0]	OK
Limit	3:	0.00	[0]	OK
Limit	4:	0.00	[0]	OK

• Assign the disabling input of the speed limit to DI[20]. Set following system variables. \$DCSS_CLLB[1].\$DITYP_HISPD = 1 (Type of input DI:1, F:35) \$DCSS_CLLB[1].\$DIIDX_HISPD = 20 (Index of input)

When the contact stop is disabled, the collaborative robot does not stop even though the external force exceeds the limit, and a serious personal injury could result. If the robot system is designed to be able to disable the contact stop, adequate risk assessment for the whole robot system is necessary to verify that the contact stop can be disabled.

7.3 LED AND BUTTON ON ROBOT ARM

The collaborative robot has the following LED and button connected to RO and RI.



By default, these LED and button are not assigned to any function. You can use these LED and Button for general purpose. The following is the example to set up the LED and Button.

SAMPLE CONFIGURATION

SPECIFICATION

- Green LED turns ON when program is running. (It keeps ON when a program is resumed by the auto resume function.)
- Red LED turns ON when an alarm occurs.
- White LED turns ON when the robot can move by pressing the button. (Remote, Program is not running, Alarm does not occur and the status of the contact stop is SAFE)

Setting

Flag and the back ground logic is used for this setting. For more detail procedure to set Flag and the back ground logic, refer to OPERATOR'S MANUAL (Basic Operation) (B-83284), Chapter of UTILITY, Section of BACKGROUND LOGIC.

- Set F[101-118] are assigned to UI[1-18]. So F[106] is assigned to UI[6:Start]
- Assign F[120] to SAFE state output. Refer to Section 2.3.
- Assign F[121] to Program running output. Refer to Section 2.3.
- Execute the following program as the background logic.

1: RO[1]= F[121]

- 2: RO[2]=(SO[3:Fault LED])
- 3: RO[3]=(SO[0:Remote LED] AND !RO[1] AND !RO[2] AND F[120])
- 4: RO[4]=(RO[3])
- 5: F[106]=(RO[3] AND (RI[1] OR RI[2]))

If the button on the robot arm is used to start robot program, emergency stop button must be prepared near the robot. The emergency stop button on the operator panel of the robot controller can be used for this purpose depending to the position of the robot controller.

7.4 OPERATION LOG BOOK

Collaborative robot state, event of payload change and so on can be recorded in the Log Book by operation log book function.

To use function, Operation Log Book option (J695) is required.

Refer to Chapter 12 of Optional Function OPERATOR'S MANUAL (B-83284EN-2) for detail of this function.

The System variable of collaborative robot operation logbook number is \$LOGBOOK.\$LOG_CLLB. Default value is 4. Logbook is started to record after first payload confirmation.

Following events are recorded to Log Book.

Message	Event	Reference
Collaborative state %d (Ch:%d)	This event is recorded when the collaborative robot state is changed.	Refer to Section 2.3 OUTPUT SIGNAL OF COLLABORATIVE ROBOT STATUS for detail.
Force limit is changed to %d	This event is recorded when the force limit is changed.	Refer to Section 2.2 SETTING COLLABORATIVE ROBOT FUNCTION for detail.
Contact stop is disabled	This event is recorded when all of force limit is disabled.	Refer to Section 2.2 SETTING COLLABORATIVE ROBOT FUNCTION for detail.
Payload number is changed to %d	This event is recorded when payload number is changed.	Refer to Section 2.2 SETTING COLLABORATIVE ROBOT FUNCTION for detail.
Contact stop(%s,%d)	This event is recorded when the robot is stopped by contact stop function.	Refer to Section 4.1 CONTACT STOP FUNCTION for detail.
Auto-resume done	This event is recorded when the program is resumed by auto resume function.	Refer to Section 7.1 AUTO RESUME FUNCTION for detail.
Auto resume canceled (%d)	This event is recorded when the program could not resumed by auto resume function. %d is the factor of this event. The factor is recorded in 2nd lines in message.	Refer to Section 7.1 AUTO RESUME FUNCTION for detail.
Reverse motion	This event is recorded when the reverse motion is executed after contact stop.	Refer to Section 4.3 RETREAT AFTER CONTACT STOP for detail.
Contact force is not released	This event is recorded when resumed or started in spite of exceeded the force limit and the collaborative state is stop.	

7.5 Variable Payload Compensation function

This function is available on the software version 7DC3/33 or after.

When a payload acts on Collaborative Robot, "SYST-325 Payload error is detected" is posted, and the robot stops. For example, in case of that a cable is attached to the robot, when the robot moves and the attitude changes, tension by cable might be detected as payload, and the robot stops.



This function cancels an external payload as such system. Firstly, move the robot in RECORD mode, and Payload is recorded at constant intervals.



During normal operation, recorded payload is canceled from payload detected by sensor.



7.5.1 How to use Variable Payload Compensation Function

Enable the function

In the collaborative robot screen, you can enable this function by "Use Payload Comp".

- After change, you need to apply DCS parameters.
- When it is "ENABLE", you can record or playback the payload.
- When you change from "ENABLE" to "DISABLE", the recorded data is not deleted.

DCS	5					
	Collabo	rative ro	bot			
	Payload	change d	istance (1	mm)		
		Current	+Limit	-Limit		
	X:	0.00	0.00	0.00	OK	
	Y:	0.00	0.00	0.00	OK	
	z:	0.00	0.00	0.00	OK	
	Rotat	ion: Curr	ent L	imit		
		0.00	[deg] 0	.20[deg]	OK	
	Enablin	g input:	ON [0]	OK	
	Passwor	d for CON	FIRM: E	NABLE	OK	
	CONFIRM	by DI:	D	ISABLE	OK	
	CONFI	RM input:	D	I[0]		
	Paylo	ad No. out	put: G	0]		
	Speed 1	imit: 250	0.00 [mm/s	5]	OK	
	Use Pay	load Comp	: 12	NABLE	OK	
	[TYPE]	CONFIRM	PEAKCLR		UNDO	

Procedure for use

- 1 Call Karel program "FCMPSTRT" at the start position of compensation.
 - 1st parameter : 1 = RECORD mode, 2 = RECORD mode (RECORD clear), 0 = PLAY mode
 - 2nd parameter : Schedule number
 - 3rd parameter : Interval for compensation (ms) Recordable time is "Interval x 1000". Minimum of interval is 8(ms).
- 2 Call Karel program "FCMPEND" at the end position of compensation.

1: J P[1] 100% FINE
2: CALL FCMPSTRT(0, 20, 100)
3: J P[2] 100% FINE
4: L P[3] 250mm/sec FINE
5: L P[4] 250mm/sec FINE
6: CALL FCMPEND
7: J P[5] 30% FINE

- 3 Execute with RECORD mode (1st parameter is 1)
 - Payload is recorded at set intervals.
 - When the recording starts, "SYST-352 Payload Record Start" warning is posted.
 - When the recording ends, "SYST-353 Payload Record Finish" warning is posted. It may take time for save recorded data to the memory. If the controller power is off during saving data, recorded data is gone.
 - When the recording fails, "SYST-354 Payload Record Failed" warning is posted. The program execution does not stop. For the cause of fail, refer to Subsection 7.5.2.
 - When you execute again with the schedule already recorded, average of payload is recorded. When you record 10 times, average value of 10 times is recorded. If you want to delete records, execute with RECORD CLEAR (1st parameter is 2).

- 4 When you execute with PLAY mode (1st parameter is 0)
 - Payload is compensated with the recorded payload.
 - When you specify unrecorded schedule, the warning "SYST-355 Payload is NOT recorded" will be posted.
 - When the compensation is failed, the warning "SYST-356 Payload Compensation is stopped" will be posted, and compensation will be zero. For the cause of fail, refer to Subsection 7.5.2.
 - In PLAY mode, specification of 3rd parameter (interval) for compensation has no meaning. Compensation works at interval at the time of recording.

Compensation Data

- The compensation data is saved below FR:¥CLLB¥
- FR:¥CLLB is saved as CLLB.ZIP in backup of "all of above".

Compensation value monitoring

Each elements of the compensation value is displayed in payload monitor screen. And you can set and output the values to GO or Register.

DCS	5									
	Collabo	rative ro	bot	:						
	Payload	Monitor:								
	FORCE	Resulta	nt	х	:	Y		z		
	Curr:	0%		0	8	0	8	C)%	
	Comp:			09	8	0	8	0	8	
	Peak:	0%		0	8	0	8	c)%	
	Output	t:								
	Curr	GO[0]	GO[0]	GO [0]	GO [0]	
	Comp	:		GO[0]	GO [0]	GO [0]	
	Warniı	ng: 50	8							
	MOMENT	Result	ant	2	х	З		2	Z	
	Curr:	0%		0	8	0	%	C)%	
	Comp:			09	8	0	8	0	8	
	Peak:	0%		0	8	0	8	C)%	
	Output	t:								
	Curr	GO[0]	GO [0]	GO [0]	GO [0]	
	Comp	:		GO[0]	GO [0]	GO [0]	
	Warniı	ng: 50	8							
	Warning	g in DSBL	:			DISA	BLE			
	[TYPE]	CONFIRM	PI	EAKCL	R			UN	DO	

7.5.2 Restrictions of Variable Payload Compensation Function

General

- This function works well in the case of that the same payload acts every time.
- When this function is enabled, Push to escape function is disabled.
- Payload is recorded in association with the point of motion path of the robot. There is no problem if the override is changed or the robot stops with WAIT statement in both RECORD mode and PLAY mode.
- In below situation, the difference between the recorded payload and actual payload, and the robot might stop by "SYST-325 Payload error is detected" alarm.
 - In the system that the payload changes every time
 - In the case of that the motion path or motion line is changed, the followings are examples
 - The program includes branches with IF and so on.
 - Execution with STEP mode
 - Addition of motion lines after recording
 - Re-teaching after recording
 - When the speed is changed, the payload is also changed.
- In the system that the external force is detected by floor vibration, vibration is changed between recording and playback, and this function will not compensates well.
- The shorter the interval for compensation, the higher the accuracy and the shorter the recording time. Recommended interval is the value obtained by dividing the operating time by 1000.

RECORD mode

• During recording, the compensation does not work. In the case of that the contact stop occurs on the way, recording fails. Set disabling input of force limit to ON, record with the contact stop disable.

When the contact stop is disabled, the collaborative robot does not stop even though the external force exceeds the limit, and a serious personal injury could result. If the robot system is designed to be able to disable the contact stop, adequate risk assessment for the whole robot system is necessary to verify that the contact stop can be disabled.

• In following cases, the warning "SYST-354 Payload Record Failed (No.X, Y)" occurs. During first recording, it becomes an unrecorded state. During after second recording, it becomes the last recorded state. X of warning message means the schedule number, and Y means the error code. Refer below causes and remedies.

Error Code	Cause	Remedy
1	Hold occurs.	Don't hold during recording.
2	Hold occurs during after second recording.	Don't hold during recording.
3	The program is aborted.	Don't abort the program during recording.
4	The program is aborted during after second recording.	Don't abort the program during recording.
5	Backward execution occurs.	Don't execute backward during recording.
6	Backward execution occurs during after second recording.	Don't execute backward during recording.
7	FCMPSTRT is called.	Don't call other FCMPSTRT between FCMPSTRT and FCMPEND.
8	FCMPSTRT is called during after second recording.	Don't call other FCMPSTRT between FCMPSTRT and FCMPEND.
9	The recording data is abnormal.	Repower the controller. If the problem is not cleared, save the diagnostic log just after failure and send the files to FANUC sales person.
10	The recoding data is abnormal.	Repower the controller. If the problem is not cleared, save the diagnostic log just after failure and send the files to FANUC sales person.
11	Payload is Unrecorded during after second recording	Execute with 1 st parameter 2 (RECORD(RECORD CLEAR))
12	The recording data is abnormal during after second recording.	Repower the controller. If the problem is not cleared, save the diagnostic log just after failure and send the files to FANUC sales person.
13	Cause Recording time exceeds.	Recording time limit is "Interval x 1000". Set the interval to longer value.

• When the controller power off occurs during recording, the recorded date does not remain.

PLAY mode

- When the program is paused, the compensation remains the value at stop.
- When the program is restarted with original path resume enable, the compensation remains the value at stop until the robot returns stop positon.
- In following cases, "SYST-355 Payload Compensation is stopped (No. X, Y)" warning occurs, and the compensation is finished. The compensation value is to zero. X of warning message means the schedule number, and Y means the error code. Refer below causes and remedies.

Error Code	Cause	Remedy
1	The program is aborted.	Don't abort the program during compensation.
2	Backward execution occurs.	Don't execute backward during compensation.
3	FCMPSTRT is called.	Don't call other FCMPSTRT between FCMPSTRT and FCMPEND.
4	The robot restarts on different motion path.	When the motion path is changed after resume, the compensation is stopped. If the original resume function is disabled, enable it. Don't change the program line
5	The recording data is abnormal.	Repower the controller. If the problem is not cleared, save the diagnostic log just after failure and send the files to FANUC sales person.

• When the controller power off occurs during playback, the compensation is finished.

7.6 DOUBLE TAP FUNCTION

When you push the parts shown in the figure below twice, the digital signal is output. For example, you set that the start signal is raised when the double tap signal is output, and then you can restart the robot with double tap after the contact stop.



The double tap function is disabled by default. If you set this function, adequate risk assessment for the whole robot system is necessary to verify that the double tap function does not cause any dangerous situation.

Condition

Only support for CR-4iA, CR-7iA and CR-7iA/L. CR-35iA is not supported.

Setting up

You can set up the double tap function with below system variables.

\$DCSS_CLLB[1].\$DOTYP_TAP: Set the type of output (DO:2, RO:9, F:35) \$DCSS_CLLB[1].\$DOIDX_TAP: Set the index of output

8

*i*HMI FOR COLLABORATIVE ROBOT

*i*HMI is a user interface with a design that even who have never used FANUC product can start up easily. Refer to OPERATOR'S MANUAL (Basic Operation) (B-83284EN), Chapter of *i*HMI. This chapter introduces functions and operation of *i*HMI for Collaborative robot.

Contents of this chapter 8.1 OVERVIEW OF THE HOME SCREEN 8.2 OPERATIONS RELATED TO SETUP 8.3 OPERATIONS RELATED TO TEACH 8.4 COLLABORATIVE ROBOT MONITOR

*i*HMI can only be used with R-30*i*B Plus/R-30*i*B Mate Plus controller.

8.1 OVERVIEW OF THE HOME SCREEN

This section explains the overview of the home screen.

For *i*HMI for collaborative robot, added functions for collaborative robots to groups of "SETUP", "TEACH", and "RUN".



Fig. 8.1 Home screen

Туре	Icon	Description
Setup	BASIC	Display the Basic Setup Selection screen. On the selection screen, you select the Initial Setup screen and End of Arm Tool Setup Screen, Collaborative Setup screen. The collaborative setting guide will guide settings that make simple pick & place programs executable and settings that enable to switch between fast modes.
Teach	CREATE PROGRAM	Display the Create a Program screen. On the Create a Program screen, you create programs using template programs (sample programs) for achieving certain objectives. You can also select a "Blank Template" for creating a program from scratch.
Run	COLLAVORATIVE	Display the Collaborative Monitor screen. Graphically displays the external force applied to the robot.

Table 8.1 Icons on the home screen

8.2 OPERATIONS RELATED TO SETUP

Tap the [BASIC] icons on the home screen and display the Basic Setup Selection screen.

On the Basic Setup Selection screen, select either the Initial Setup screen or the End of Arm Tool Setup screen, Collaborative Setup screen.

The screen display flow for setup is shown below.



Fig. 8.2 Screen display flow (Setup)

For Initial Setup Guide and End of Arm Tool Setup Guide, refer to OPERATOR'S MANUAL (Basic Operation) (B-83284EN), Section of Operations related to Setup.

8.2.1 Collaborative Setup Guide

The collaborative setup guide will guide settings that make simple pick & place programs executable and settings that enable to switch between fast modes. The collaborative setup guide screen is shown below.



Fig.8.2.1 Collaborative Setup Guide screen

The main setup items on the Collaborative Setup Guide are as follows:

- Payload Setup
- Contact-Stop Invalid Area (Payload change distance)
- Tool Setup
- High Speed Mode

NOTE

- 1 When the Collaborative Setup guide is completed, the tool setting guide is also automatically set to the completion state.
- 2 When the Collaborative Setup guide is completed, go to DCS screen.

8.3 OPERATIONS RELATED TO TEACH

Tap the [Create Program] icon on the home screen and display the Create a Program screen.



The screen display flow when there are no programs is shown below.

Fig.8.3 Screen display flow (when there are no programs)

Refer to OPERATOR'S MANUAL (Basic Operation) (B-83284EN), Section of Operations related to teach.

8.3.1 Create Program

On the Create a Program screen, you create programs using template programs (sample programs) for achieving certain objectives. You can also select a "Blank Template" for creating a program from scratch. The Create Program guide screen is shown below.



Fig.8.3.1 Create a Program Screen

The templates that can be selected on the Create Program Guide screen are as follows:

- Blank Template
- Pick & Place with light work
- Pick & Place with heavy work

8.3.2 Pick & Place with Light Work

This is a template for creating pick & place program with workpiece weight less than 1 kg. You can create a pick & place program with only three teachings.

Teaching points that require teaching by Operator are 1st, 2nd, and 5th point. Other teaching points are taught automatically from 1st and 5th point.



Fig. 8.3.2 Create a Pick & Place with light work screen

If the workpiece is less than 1 kg, you can usually operate without stopping the robot without switching payload.

NOTE

If the workpiece is less than 1 kg, the robot may stop. In that case, please use Pick & Place with heavy work template.

8.3.3 Pick & Place with Heavy Work

This is a template for creating a pick & place program with a workpiece weight of 1 kg or more. You can create a pick & place program with only 5 teachings.

Teaching points that require teaching by Operator are 1st, 2nd, 3rd, 4th and 7th point. Other teaching points are taught automatically from 1st and 7th point.



Fig. 8.3.3 Create a Pick & Place with heavy work screen

If the workpiece weight is 1 kg or more, the robot stops unless the payload switching is properly performed. Please make appropriate settings before operating.

If switch payload, Contact Stop function is disabled In Contact-Stop Invalid Area. The robot does not stop, if an operator contacts. If switch payload, adequate risk assessment for the whole robot system is necessary to verify that the contact stop function becomes invalid within the Contact-Stop Invalid Area.

8.4 **Collaborative Monitor**

Collaborative Robot Monitor provides the screen that can be graphically monitored external forces on cooperating robots.

Tap the "COLLABORATIVE" icon on the home screen to display the Collaborative Robot Monitor screen.



HOME screen

Collaborative Robot Monitor Screen

Fig. 8.4 Screen display flow (Collaborative Robot Monitor)

8.4.1 **Collaborative Robot Monitor Screen**

Collaborative Robot Monitor Screen shows the following information.

- (1) Collaborative Robot
- (2) Contact Stop status
- (3) The value of External Force, Payload and Moment
- (4) Active Payload Number
- (5) Direction and magnitude of the Force



Fig. 8.4.1 Information on Collaborative Robot Monitor screen

Followings are explanation of the information on the screen.

Collaborative Robot (1)

Collaborative robot graphic is shown. The robot model is the same posture as the actual robot.

Contact Stop status (2)

This parameter is the status of contact stop status of the collaborative robot. The contact stop status has the following four statuses. Depending on each state, the background color of the status also changes. Please refer to "2.3 OUTPUT SIGNAL OF COLLABORATIVE ROBOT STATUS" for details of the parameter.

SAFE (green) : The contact stop function is enabled and the safety is confirmed.

STOP (red) : The contact stop function is enabled and the safety is not confirmed.

DSBL (yellow) : The contact stop function is disabled.

ESCP (yellow) : In Push To Escape.

The value of Force, Payload and Moment (3)

The magnitude of the force applied to the robot is displayed. Please refer to "4.2.3 Collaborative Robot Monitor Screen" for details of the parameter.

Force[%]

External force applied to the robot is displayed by a blue bar. The maximum value is assumed to be 100%, and when it exceeds 80% the color of the bar changes to red.

Payload[%]

Steady load applied to the robot is displayed by a green bar. The maximum value is assumed to be 100%, and when it exceeds 80% the color of the bar changes to red.

Moment[%]

Moment applied to the robot is displayed by a green bar. The maximum value is assumed to be 100%, and when it exceeds 80% the color of the bar changes to red.

Active Payload Number (4)

The selected payload number is displayed. If a comment is set for the selected payload, that comment is also displayed.

Direction and magnitude of the Force (5)

The direction and magnitude of the external force applied to the robot is displayed. From the inner surface of the sphere drawn to enclose the robot, a blue cone is displayed towards the direction of force applied to the robot. The length of the cone changes depending on the magnitude of the external force. When the magnitude of force applied to the robot exceeds 80 %, the color of the cone changes to red. If the magnitude of force applied to the robot is less than 20%, the cone is not displayed. Please refer to "4.2.3 Collaborative Robot Monitor Screen" for details of external force.

APPENDIX

A TROUBLESHOOTING

A.1 INDELIBLE ALARM

"SRVO-358 DCS INVPRM alarm" occurs.

Verify followings.

- The group number setting of the collaborative robot is correct.
- The force limit settings are between 0 and 150, inclusive.

"SRVO-473 DCS CLLB CC_EXTF alarm" occurs.

Verify followings.

- Refer to A.2 "SYST-320 Program paused by contact stop" or "SYST-325 Payload error is detected" occurs, even though nobody contacts the robot."
- Replace the force sensor.

"SRVO-477 Calibration data error" occurs.

Verify if backup files were loaded after initial start. If not, execute one of the following items. After executing, check that "Force Sensor Serial number on the manufacturing plate" corresponds to that of "Force Sensor Serial number on TP". Refer to PREFACE of each MECHANICAL UNIT OPERATOR'S MANUAL (B-83734EN, B-83774EN).

- Restore appropriate backup files. If necessary, perform "APPLY TO DCS PARAMETER". Refer to section 1.3 of Dual Check Safety Function OPERATOR'S MANUAL (B-83184EN).
- Copy "CLLB.CM" file from the CD-R which is appendix of the collaborative robot to a memory card or USB memory. Perform the same file with a teach pendant, set the sensor calibration parameter. After this change, perform "APPLY TO DCS PARAMETER". Refer to section 1.3 of Dual Check Safety Function OPERATOR'S MANUAL (B-83184EN).

"SRVO-481 Invalid payload number" occurs.

Set correct payload number except for 0.

"SYST-326 Need to confirm payload" occurs.

Payload confirmation operation is necessary after power on. Refer to Chapter 3.

A.2 STOP DURING MOVE

"SYST-320 Program paused by contact stop" or "SYST-325 Payload error is detected" occurs, even though nobody contacts the robot.

Verify followings.

- The robot does not touch any object.
- Current payload setting is consistent with the actual payload.
- Robot installation plate is fixed.
- Robot installation plate is not warped. When Robot installation plate is warped, correct warped installation surface using spacer etc.
- Vibration of the floor or the hand is not applied to the robot. If a vibration is applied to the robot, deal so that the vibration is not applied to the robot. Reducing the speed or extending the acceleration time with Acceleration override instruction may reduce the vibration of the hand.
- "Force Sensor Serial number on the manufacturing plate" corresponds to that of "Force Sensor Serial number on TP". Refer to PREFACE of each MECHANICAL UNIT OPERATOR'S MANUAL (B-83734EN, B-83774EN).

"SYST-322 Auto status check time out" occurs.

External force reset must be performed periodically. Refer to the subsection 4.2.1.

A.3 STOP DURING PERIODICAL CHECK

Refer to Chapter 6.
INDEX

<A>

ABSTRACT	1
AUTO RESUME FUNCTION	7
Auto Status Check16	5

<C>

Collaborative Monitor	54
COLLABORATIVE ROBOT MONITOR	16
Collaborative Robot Monitor Screen	
Collaborative Setup Guide	
CONTACT STOP FUNCTION	15
Contact Stop in Teach Mode	16
Create Program	51

<D>

DISABLING CONTACT STOP IN PROGRAM .	28
DOUBLE TAP FUNCTION	47

<H>

HIGH SPEED MODE	38
How to use Variable Payload Compensation Function.	44

</>

<i>i</i> HMI FOR COLLABORATIVE ROBOT	48
INDELIBLE ALARM	59

<L>

LED AND BUTTON ON ROBOT ARM	41
Limiting Picking Area	24

<0>

OPERATION LOG BOOK	42
OPERATIONS RELATED TO SETUP	49
OPERATIONS RELATED TO TEACH	51
OUTPUT SIGNAL OF COLLABORATIVE ROBOT	
STATUS	11
OVERVIEW OF THE HOME SCREEN	48

<P>

PAYLOAD CONFIRMATION OPERATION AFT	ΈR
POWER ON	13
Payload Confirmation without Teach Pendant	14
Payload Monitor	17
PERIODICAL CHECK	35
Pick & Place with Heavy Work	53
Pick & Place with Light Work	52
PICKING WORKPIECE	21
PREFACE	p-1
PROGRAMMING	21
PUSH TO ESCAPE	20

<R>

RESTRICTION OF COLLABORATIVE ROBOT	
Restrictions of Variable Payload Compensation	
Function45	;

RETREAT AFTER CONTACT STOP......19

<S>

SAFETY PRECAUTIONS	s-1
SETTING COLLABORATIVE ROBOT FUNCTION	۶4
SETTING SAFE I/O DEVICE	3
SETTING UP	3
SETUP STEPS	2
Simple Picking Program	21
SOFTWARE OPTION FOR COLLABORATIVE	
ROBOT	1
SPECIFICATION OF COLLABORATIVE	
FUNCTION	15
STOP DURING MOVE	60
STOP DURING PERIODICAL CHECK	60

<T>

TROUBLESHOOTING59

<U>

UTILITY	7
---------	---

<V>

Variable Payload Compensation function	43
Verify Payload Setup	9

REVISION RECORD

Edition	Date	Contents
02	Oct., 2017	Support small collaborative robot, Add monitor function, Add Variable Payload Compensation function, Add Double Tap function, Add <i>i</i> HMI
01	Sep., 2015	



